Stockholm School of Economics

Master Thesis in Finance December 2018

Systemic risk and the market-to-book value of banks' equity

Florian Meier

41338@student.hhs.se

Yuqing Meng 41253@student.hhs.se

Abstract

The goal of this thesis is to investigate whether market participants considered possible government guarantees when valuing banks' equity before the recent financial crisis and to what extent the market value of banks' equity is useful to estimate their exposure to systemic risk. Using a theoretical approach, we first present our argument that anticipated government guarantees in case of a systemic breakdown lead to financing costs of debt being independent from a bank's exposure to systemic risk. Since debtholders classify their debt repayments as almost riskless due to expected government intervention ("government guarantees") in case of a default in a crisis, they require only little compensation for their contribution. In return, banks increase their exposure to systemic risk to earn additional future cash flows, leading to a higher market-to-book value of their equity.

We test this argument by regressing individual banks' average losses on the worst days during the crisis (as a proxy for their systemic risk exposure) on their market-to-book value of equity before the crisis hit. We find a highly significant positive relationship between banks' market value of equity and their exposure to systemic risk. This leads us to the conclusion that first, market participants priced anticipated government guarantees in banks' market value of equity and second, the market-to-book value of a bank is an indicator for its exposure to systemic risk. Several robustness tests support our findings.

Keywords: Banks, Financial Crisis, Government Guarantees, Market-to-book Value, Risk

Acknowledgements: We would like to thank our supervisor, Assistant Professor Adrien d'Avernas, for his valuable input and guidance throughout our thesis writing process.

Table of Contents

1.	Intr	oduction
2.	Lite	erature review
2.	1	Systemic risk and tail risk in the financial system
2.2	2	Measurement of tail risk
2.	3	Measurement of government guarantees
3.	The	oretical model
3.	1	Impact of government guarantees on the financial system's risk-taking behavior13
3.2	2	Impact of government guarantees on banks' MBV16
3.	3	Comparative statics
3.4	4	Hypotheses development
4.	Res	earch design and methodology26
4.	1	Research design
4.	2	Data and methodology27
5.	Fine	dings
5.	1	Main regressions
5.2	2	Robustness tests
5.	3	Estimation of tail risk
6.	Dis	cussion and conclusion
7.	Ref	erences
8.	App	pendix

1. Introduction

The collapse of the US investment bank Lehman Brothers in 2008 marked the peak of a global financial crisis that is considered as the worst economic downturn since the Great Depression. What started with distressed subprime mortgages that seemed to mainly affect the American housing market quickly spilled over to the whole financial industry in the US and eventually, in the whole world. What made this crisis considerably severe is that it did not only affect banks and financial institutions, which were directly involved in the business, but led to a global economic downturn, affecting real growth in different industries. The large negative implications of a possible further breakdown of the world economy led to many governments saving banks in their respective countries from bankruptcy (U.S. Department Of The Treasury, 2012).

The financial crisis of 2008 showed in a dramatic way how strong the different financial institutions in the world are interrelated with each other and the rest of the economy. Especially in the US, where most of the banks saved by a bailout operate, a long public discussion evolved around the question: Why do taxpayers have to save exactly these institutions that were responsible for the crisis in the first place? The answer seems straightforward: The collapse of Lehman Brothers has shown that further failure of institutions that are too big and too interconnected with the rest of the financial industry can cause an even worse downturn, not only for the rest of the industry, but for the whole economy. Many institutions have been saved because they were simply "too big to fail".

In this thesis, we would like to investigate whether the financial system had been aware of this phenomenon before the actual bailouts happened. Did market participants consider possible government guarantees when valuing banks' market values and, if yes, does this help us to obtain information about the current exposure of certain institutions to systemic risk?

We explore this question by looking closer at banks' market-to-book values (MBV) in 2006, the year before the global financial crisis. Using a simple theoretical framework, derived by Atkeson et al (2018), we argue that the MBV can be interpreted as a risk factor, given anticipated government guarantees are priced in the market value of equity.

Government bailouts usually come into play when a bank's losses wipe out all its equity, leading to a state where the institution is unable to pay back its obligations to debtholders. In this case, governments can support them with taxpayers' money to promise for banks' liquidity.

Such bailouts usually happen if a bank's failure to meet its obligations is expected to cause severe additional problems, which is the case in a financial crisis when several institutions at the same time encounter losses. Such guarantees are usually not announced explicitly, but they might be expected by the financial markets and, hence, implicitly priced in banks' market values of equity. An important measure when talking about government bailouts is, hence, banks' potential losses during severe economic breakdowns, called their "tail risk". If the financial system expects this tail risk to be compensated by taxpayers in case of a crisis, cost of debt becomes independent from such risks. Because high risks come with high returns, banks can make additional profits through loading on tail risk without having to pay additional cost for their leverage since debtholders see their contribution as riskless. This leads to rising cash flows, which are priced in the market value of equity of banks.

We test this theory in an empirical study by regressing tail risk (proxied through the average returns of banks on the worst days of the financial crisis) on the market-to-book value of banks in 2006, the year prior to the crisis. This approach is closely related to the work of Meiselman et al (2018). The results show a highly significant correlation between banks' MBV before the crisis and their losses during the crisis. Market-to-book values of equity can explain almost 30% of the variation of banks' tail risk. This leads us to two main conclusions: First, the results indicate that anticipated government guarantees were priced in the market values of banks' equity in 2006. Second, the size of MBV of a bank contains information about its tail risk and might serve as an estimate for a measure of risk in different times. The latter one, however, requires a more cautious formulation: Even though we find MBV to be a significant factor in explaining our measure for tail risk, this does not necessarily mean that MBV is the most important or only important variable in explaining the risk measure. Using several control variables, such as size or leverage, we can show that the explanatory power of MBV to describe tail risk does not stem from its correlation with other values. However, we can also confirm a highly significant correlation of factors other than MBV with banks' exposure to systemic risk.

By applying these findings, we develop a multivariate model to estimate a current tail risk measure in an out-of-sample application. We show that, by looking at MBV and other variables from banks' financial reports, we can explain more than 60% of the variation of our measure for banks' tail risk during the financial crisis. Using the same model to estimate tail risks in 2016, we find that little has changed regarding the total level of expected losses in case of another systemic breakdown compared to 2006. What is interesting is that at the same time, average MBV of banks decreased. This shows that even though market-to-book value seems

to be a good indicator for tail risk, other factors must be considered as well when estimating banks' exposure to systemic breakdown.

The remainder of this thesis is structured as follows: Section 2 introduces the background of our research topic and presents the relevant literature in this area. In section 3, we present a theoretical model further describing the connection between government guarantees, tail risk and MBV. Section 4 gives a detailed overview of the design of our empirical analysis and the data we use. Section 5 provides the results and section 6 concludes.

2. Literature review

The main goal of this thesis is to investigate the relationship between a bank's market-to-book value and its risk-taking behavior. To present our reasoning, in this section we introduce banks' characteristics and the potential threat to the economy in a crisis, which lead us to focus on a special kind of risk, systemic risk. In the second step we mention common methods on observing as well as measuring systemic risk. Finally, we give a short overview over other factors influencing a bank's risk exposure.

2.1 Systemic risk and tail risk in the financial system

The start of the global financial crisis through the bankruptcy of Lehman Brothers, described shortly in the introduction, showed how detrimental a failure of a single bank can be for the whole global economy. Furthermore, such a single "systemic event", can lead to the breakdown of other banks. We call this risk where several financial institutions or markets at the same time encounter failure "systemic risk". Systemic risk imposes two main problems for the economy: First, it is costly for economies since a breakdown of the financial sector can lead to a liquidity shortage due to an undercapitalization and hence a hampering of economic growth in general. Negative spillover effects, such as negative performance on bank-dependent firms after a systemic crisis, are noted in literature (Chava & Purnanandam, 2011). Second, potential bailouts of systemically important banks lead to increased cost for taxpayers.

Due to their important role of providing financing to the economy, their high levels of leverage, and their interconnectedness, banks can be especially vulnerable to systemic risk (De Bandt & Hartmann, 2000). This means that they might encounter strong decreases in return or liquidity when other parts of the financial system or market fail.

The high amount of leverage in banks is one reason why financial firms are especially vulnerable in times of a severe systemic crisis, although a high amount of leverage is common in the financial sector and does not automatically imply higher financial stress (DeAngelo & Stulz, 2013). By being financial intermediaries, banks lend out money from creditors to different debtors and perform a maturity transformation, hence turning short term deposits into long term credits. In order to guarantee liquidity, banks must diversify the risk from their debtors by lending out to uncorrelated industrial and private sector participants, which leads them to have lower risk of operational leverage or less industry-specific or idiosyncratic risk. However, financial intermediaries have little methods to differentiate their line of services, and

in a competitive setting, they cannot choose the spreads freely, which determine their return rates (Saunders & Walter, 1994). In order to remain profitable to their equity shareholders, banks have to increase their return on equity, a task mainly done by taking on high leverage, which can be paid off in normal times since operational risk is low.

Another idiosyncrasy with banks' business model is that due to maturity transformation banks are reliant on debtors' trust. A single trivial event, i.e. "fake news", could lead to a drop in creditworthiness of an institution and to accumulated deposit withdrawals which could ultimately result in illiquidity of a bank. In other words, banks are not only dependent on their operational business decisions by choosing the right projects and creditors, but also on the beliefs of their debtors regarding general safety and the possibility of bank runs (De Bandt & Hartmann, 2000).

Furthermore, the interconnection between financial institutions through the interbank money market and payment security settling systems, not at last increased through various financial innovations for securitization and other credit risk transfer, leads to a system of contagion. In a systemic crisis, the failure of one bank to meet its obligations can have a tremendous effect on the liquidity of its creditor banks, which could amplify in a "domino" effect. Hence, even though in normal times, banks are relatively safe since their operational risk is very low, they can suffer huge losses during crisis times when several banks experience losses at the same time (De Bandt & Hartmann, 2000).

The high importance of this systemic risk for the financial industry calls for a bank-specific risk measure that focuses on the exposure of an individual bank to a severe systemic crisis. We call this risk "tail risk". Tail risk, hence, reflects the potential loss of a bank in extremely bad times, which is uncertain in normal times. Systemic risk and tail risk are, therefore, closely related. While systemic risk describes the risks for the whole financial system coming from a systemic breakdown, tail risk describes the effects of such an event on one certain bank. The measure is different from traditional risk measures such as the co-movement of returns with the market return, which do not mainly focus on crisis times. The most popular measure, the CAPM-beta (measuring the so called "systematic risk"), for example does not focus on systemic risk since it is mainly focused on the co-movement of banks with the market in normal times rather than on rare systemic downturns. An example is its inability to capture the higher risk during the burst of the housing bubble (Caporale, 2012; Adrian et al, 2015).

The distinct characteristics of banks' risk exposure is the reason for many researchers omitting this sector when making general regressions for risk and return. For example, Fama and French (1992) exclude the financial sector before building their portfolios, and the Fama-French-5-factor model does not seem to explain average returns in the financial sector (Adrian et al, 2015).

Hence, we have to focus on other ways to measure bank risk, which focus on tail risk. In the next section, we describe ways to describe and measure banks' tail risk.

2.2 Measurement of tail risk

We have shown in the last section that banks have relatively low operational risk and, hence, relatively low risk in normal times. Therefore, we focus on the losses of banks in extremely bad times, on their tail risk. Since this risk is not observable, different measures for risk estimation and prediction are mentioned in literature. We will present several ways to estimate such tail risk per bank and explain why the MBV could be a potential explaining factor for tail risk. Before doing so, however, we have to find a way to measure actual tail risk in order to be able to test the explanatory power of different variables. Because tail risk focuses on economic downturns, we can only measure it with sufficient accuracy ex post, after the financial system has experienced a severe crisis.

2.2.1 Ex post measures

The goal of this thesis is to investigate whether MBV can be used to better understand government guarantees and tail risk of banks. In order to do so, we first need a way to measure this tail risk based on real market events. Acharya et al (2016b) construct a simple metric related to systemic risk, which describes the losses of firm on the days with the worst performance of the whole market, the marginal expected shortfall (MES). Furthermore, they argue that this measure is closely related to each firm's contribution to a systemic crisis, the systemic expected shortfall (SES). MES per bank is theoretically defined as follows:

$$\frac{\partial ES_{\alpha}}{\partial y_i} = -E[r_i | R \le VaR_{\alpha}] \equiv MES_i$$

The sensitivity of the expected shortfall of the overall firm, consisting of different risk factors y, to a specific risk factor y_i , is the marginal expected shortfall of this factor.

In other words, the measure tells us how much an individual bank loses, given the aggregate market return falls below a certain level, here called value-at-risk (VaR) (see also section

2.2.2). The researchers provide an easy but very effective approach of measuring MES: MES is the return of a single bank on the 5% worst days of the market over a certain period, i.e. the return of an individual bank during extremely hard (crisis) times. Hence, MES is an easy way to measure the tail risk of a bank, which we have defined as the performance of an individual institution during extremely bad systemic downturns.

The measure, however, can only be calculated after a crisis has occurred. It can, hence, be used to look back in time and to say something about a bank's tail risk just before a certain crisis, using the observations of this bank during the crisis. It is, however, not an estimate of current, actual tail risk.

2.2.2 Ex ante measures

MES as a measure for tail risk is a simple way to say something about the past. The challenge, however, is to estimate tail risk ex ante, hence, before a crisis occurs. This field of research can be divided roughly in three categories: First, there is a lot of literature focusing on the interdependencies of the whole financial system and how the current state of the system can be measured. These measures, however, do not mainly focus on individual banks. Second, model-based estimation techniques can be applied to look at possible reactions of a bank's portfolio to a systemic crisis. And third, certain variables based on banks' book or market values can be identified, which might be used as proxies for tail risk due to their correlation with the mentioned tail risk measures. We will shortly describe all three categories.

One focus of research is to describe the interdependences of individual banks with the whole system during a crisis in general. Such theories are focused on the sources of a crisis and the interconnectedness of the banking sector, which can lead to an amplified exposure and "domino-effects". One example is the Δ CoVaR by Adrian and Brunnermeier (2016), which integrates spillover effects, by measuring the contribution of an institution *i*, such as an individual bank, to the conditional VaR of another institution *j*, such as the whole financial system, given a systemic event. Another approach is by Weiß et al (2014), who apply the measure of lower tail dependence (LTD) which is the probability of the stock returns of an individual bank and the market being in the lower tail. Moreover, Patro et al (2013) identify that stock return correlations among the 22 largest bank holding companies increase with significant market events, and hence can explain systemic risk. They further decompose risk into systematic and idiosyncratic risk and find that recent increases in systemic risk can be

attributed towards an increase in correlation of banks' idiosyncratic risk, which confirms the thesis that higher bank interdependence leads to higher systemic risk.

However, models describing mechanisms in the whole system are not useful to describe risk exposures of individual banks. Looking at single institutions, we want to find a way to answer the question: "How would a certain bank react tomorrow if a crisis occurred and the whole financial system would break down, given markets are in a normal state today?" This question is especially important for regulators and supervisory-institutions who try to assess the riskiness of banks for the whole financial system. Such institutions usually use model-based approaches to estimate tail risk. By the Basel Capital Accord II (Basel Committee on Banking Supervision, 2003), as an internal model, banks should use the risk measure of value-at-risk. Value-at-risk (VaR) measures the minimal loss on an asset over a period given a confidence level, hence the probability of the return being smaller than the VaR should be α , such that:

$$Pr(R < -VaR_{\alpha}) = \alpha$$

VaR is a model-based measure that tries to estimate risk directly through assuming certain return distributions. Since the input variables for calculating the VaR are usually dependent on the current market value and its standard deviation, it measures the losses resulting from normal market movements (Allen & Saunders, 2004). This focus on "normal times" has been criticized and led to undercapitalized book exposures in times of the crisis (Basel Committee on Banking Supervision, 2014).

This led to a change in measurement in the Basel framework, replacing it by expected shortfall as the regulators' measurement of tail risk. The expected shortfall measures the expected loss under the condition of a higher loss than the VaR and hence the capital charge from this measure focuses more on risks resulting from really bad events for an individual institution.

$$ES_{\alpha} = -E[R|R \le -VaR_{\alpha}]$$

However, even though this measure focuses on extremely bad events, it still does not capture explicitly what happens to a single bank if all other banks experience bad times, too (Allen and Saunders, 2004). Like value at risk, the measure does not take into account the current state of the market and, hence, does not tell us anything about crisis times.

Besides their focus on normal times, model-based approaches are also criticized because they are easy to manipulate by regulated entities (Behn et al 2016). Since regulators define capital requirements, such as leverage or liquidity ratios, using these model-based approaches, banks

that face binding requirements might have a higher incentive to underreport risk (Begley et al 2017).

A third way of measuring tail risk is, hence, to find ratios, which tend to be correlated with tail risk. Such approaches do not try to model tail risk directly, but to proxy it through other measures that are less prone to manipulation.

One work in this field that is very closely related to this thesis is from Meiselman et al (2018), who argue that book returns of banks are a better proxy for tail risk than model-based measures, such as ones using risk weighted assets. They describe a financial institution as an asset portfolio, which should generate higher returns due to higher exposure to systemic risk. Through several regressions, the researchers can show that banks return on equity before the recent financial crisis can explain about 20% of the variation in their tail risk (measured through MES), which is higher than the risk proxy generated using risk weighted assets.

This thesis takes the idea of Meiselman et al (2018) and develops the reasoning further. We use the same idea of using non risk-specific measures in normal times to proxy for tail risk. Using linear regression, we investigate whether MBV of banks before the recent financial crisis would have been a good indicator for their tail risk. MBV is different to other measures that have been proposed in the way that it includes market values, while return or book leverage focus on book values. Hence, the measure includes investors' market expectations about future cash flows, whereas accounting ratios focus on historical data, such as the profitability from the previous year.

By choosing MBV as our explanatory variable we incorporate investors' beliefs in the measure, which allows us to connect tail risk to anticipated government guarantees. If anticipated government guarantees were priced by market participants, they should be reflected in a higher market value of equity of banks.

2.3 Measurement of government guarantees

Recent literature describes the effect of government guarantees on banks' behavior. Due to moral hazard between banks and government, government guarantees might distort banks' decision to take on risk (Calomiris, 1990). Since governments guarantee the payback of deposits in crisis times, these deposits are valued as safer than they would be without guarantees. As a result, banks face lower financing costs, leading them to take on higher risk (see section 3). Gropp et al (2013) show in a natural experiment, that banks removed credit risk

after explicit government guarantees were removed, a behavior not observed in the control group. Allen et al (2018), however, show that government guarantees do not only have negative distortionary effects: Deposit insurance might still be welfare improving, since it provides banks with liquidity, which lead banks to be more efficient in liquidity transformation. Further literature measures government guarantees by comparing differences in bond credit spreads of different institutions, whereas systemically important institutions are proxied by looking at their size. Acharya et al (2016a) find that the largest financial institutions are less sensitive to changes in bond credit spreads, compared to smaller financial institutions. Since this difference in sensitivity does not hold for non-financial firms, the authors conclude that this relationship can be attributed to government guarantees, which allow large financial firms to borrow at subsidized rates. Similarly, Gandhi & Lustig (2015) find out that a portfolio with a long position in a stock portfolio of the largest U.S. commercial banks and a short position in a stock portfolio of the smallest banks underperforms an equally risky portfolio of all non-bank stocks. This arbitrage opportunity implies that large banks do not have to pay as high required returns as large firms in general. The difference can be explained by the government guarantees, which apply for large institutions rather than for small ones, due to their importance for the financial system. Kelly et al (2011) compare the price of out-of-the-money index options with the price of a basket of out-of-the-money options of the individual banks. They argue that index options are more exposed to systematic risk, while the basket is more exposed to individual risk. Since systemic risk is guaranteed by the taxpayer, index options should trade at a higher price compared to the basket of individual options. They find evidence that this is the case, supporting the argument that anticipated government guarantees in case of a systemic crisis are priced in the option premium.

In the next section, we will further describe the connection between anticipated government guarantees, MBV and eventually, tail risk.

3. Theoretical model

In this section, we use a simplified version of the two-state model by Atkeson et al (2018) to illustrate that the market-to-book value can be indeed a risk factor for banks. In the model, under the assumption of government guarantees banks have an incentive to choose the risky project over the risk-free project, which leads them to loading on more risk in expectation for government bailout in a possible crisis. Additionally, we show that an increase in the risk of the project leads to a higher market-to-book value.

The model we present assumes that government guarantees influence the behavior of the whole financial industry. It explains why the participants act differently under the assumption that they are not accountable for the possible negative consequences of their behavior, leading to negative social externalities. In our model, however, not only banks face moral hazard, the whole financial system expects these implicit guarantees. Debtholders offer banks lower interest rates since they expect their debt to be bailed out in a crisis, thus inducing banks to choose risky projects. Equityholders also benefit from these government guarantees since the consequences of a breakdown do not have to be carried by any party of the financial firm, in this model, but from the taxpayers due to its spillovers to the whole economy, this creates negative social externalities for the society.

It is important to mention that here, we talk about the whole financial industry, rather than one individual bank. This representative agent model models the behavior of one bank and one financier, in order to describe how banks and debtholders would work in a financial system.

3.1 Impact of government guarantees on the financial system's risk-taking behavior

In our model, the financial industry consists of two participants. The bank (*B*) starts with a certain amount of its own money (*N*). Furthermore, it gets further resources from the debtholder, here called financier (*F*), who requires the risk-free compensation R^f for its contribution. As a debtholder, the financier has a senior claim to the equityholder, which means that the interest is paid back first. The bank invests all the money (*I*), financed by leverage ρ , such that $I \times \rho$ equals the dollar-contribution of the financier and $I \times (1 - \rho)$ equals the bank's own fraction, *N*.

Now, we assume a two-state-model. In the first state, the bank can decide between two possible projects to invest in, a risky project and a safe project. These two projects will have different outcomes in the second state, depending on which second state will occur, the crisis state or the normal state:

	safe project	risky project
normal state n	R ^f	$R^n > R^f$
crisis state c	R^{f}	$R^{c} < R^{f}$

With R^{f} being the risk-free rate and q_{c} being the risk-neutral probability of the crisis state, *c*. Furthermore, we assume perfect competition and risk-neutral probability:

$$(1) (1 - q_c) \times R^n + q_c \times R^c = R^f$$

We now focus on the financier, who requires an expected compensation of R^f for its investment, regardless of the outcome of the economy. If the bank choses to invest in the safe project, the financier gets a guaranteed dollar payoff of $I \times \rho \times R^f$. If the bank invests in the risky project and the normal state occurs, the financier will get $I \times \rho \times R^f$ too, since $R^n > R^f$, by definition. However, if the bank decides to invest in the risky project and the crisis state occurs, there are two options:

- 1. $I \times R^c > I \times \rho \times R^f$: In this case, there is no risk for the financier. Regardless of which project is chosen, it will get a safe payoff of $I \times \rho \times R^f$, even in the crisis state.
- 2. $I \times R^c < I \times \rho \times R^f$: Given the risky project is chosen, the bank is unable to pay off its obligation of $I \times \rho \times R^f$ to the financier in the crisis state and, hence, choses bankruptcy. In this case, the financier gets what is left, $I \times R^c$, which is smaller than its required compensation.

The question now is: Which project is chosen under which circumstances, given the credible condition that $R^c < \rho \times R^f$? This depends on whether there are government guarantees.

3.1.1 Decision without government guarantees

To agree to the risky project, the financier requires a higher return than R^{f} in the normal state to make sure the expected payoff still equals R^{f} . Or mathematically, the financier wants to make sure that:

(2)
$$[(1 - q_c) \times R^F + q_c \times R^c] \times I = I \times \rho \times R^f$$

or

(3)
$$R^F = \frac{\rho \times R^f - q_c \times R^c}{(1 - q_c)}$$

where R^F is the financier's payoff in the normal state. In other words, even though the financier only requires the risk-free rate, the bank has to promise more due to crisis risk. This leads to bank's expected dollar return of the risky project:

$$(4) (1 - q_c) \times (R^n - R^F) \times I$$

(Due to the derivation above, R^F is the fraction of the whole investment I that goes to the financier. The leverage ratio ρ is already included)

We use the fact that $R^F = \frac{\rho \times R^f - q_c \times R^c}{(1 - q_c)}$ and we rearrange (4) as follows:

(5)
$$(1-q_c) \times \left(R^n - \frac{\rho \times R^f - q_c \times R^c}{(1-q_c)}\right) \times I = \left((1-q_c) \times R^n - \rho \times R^f + q_c \times R^c\right) \times I$$

Knowing that, by the definition of risk neutral probability (1), we can write the banks' expected return for the risky project as:

(6)
$$(1 - \rho) \times I \times R^f$$
 or $N \times R^f$

It can be easily seen that this exactly equals the banks' expected return of the safe project. Hence, for the case that $I \times R^c < I \times \rho \times R^f$, it is never profitable to choose the risky project, assuming no government guarantees. In our model with only one representative agent, no risky projects would be executed.

Our model assumes that the financier can control the decision of the bank to choose the riskless project. Often, however, the financier has to make the decision for financing before the bank decides to invest in the risky or riskless project, i.e. a state of moral hazard, since the actual action is unobserved. In a non-repeated game, thus facing no consequences for its actions in the previous period, the bank might have an incentive to deviate, since it could earn more if it chose the risky project while paying the riskless rate for financing (see 3.1.2). The financier would expect this action and demand only R^F . Since this is a state where no one could change their action to get to a state where they are better off, this is a Nash equilibrium, in similar manner to the prisoner's dilemma. In practice, financiers have the options to exert control using covenants and banks also have to care about the financiers to obtain future financing, which makes this option less likely.

3.1.2 Decision with government guarantees

Now, we assume that there is a government, ready to guarantee that the financier will always get a return of R^f on its money, or $I \times \rho \times R^f$ in dollars, respectively. For the safe project, nothing changes in this case. Both, financier and bank get a guaranteed payoff of R^f on their contribution. Assuming that the risky project is chosen and that the normal state occurs, nothing changes either. However, in case of a crisis, the government now guarantees a payoff of $I \times \rho \times R^f$ for the financier, meaning that it closes the gap by paying out $I \times \rho \times R^f - I \times R^c$. This means that the financier does not require R^F from above anymore since its payoff will be $I \times \rho \times R^f$, even in the crisis state. The expected return for the bank, hence, becomes:

(7)
$$(1 - q_c) \times (R^n - \rho \times R^f) \times I$$

We also rearrange this equation:

risky payof
$$f_{bank,gov.guarantee} = [(1 - q_c) \times R^n - (1 - q_c) \times \rho \times R^f] \times I =$$

(8) $[(1 - q_c) \times R^n + q_c \times \rho \times R^f - \rho \times R^f] \times I =$

We know from above that the bank's payoff for the safe project equals its payoff for the risky project without government guarantees, see assumption in (1):

(9)
$$(1 - \rho) \times I \times R^f = [(1 - q_c) \times R^n + q_c \times \mathbf{R}^c - \rho \times R^f] \times I$$

The equations are almost identical and only differ in the bold terms, which enter with a positive sign. Since we know that $\rho \times R^f > R^c$ by assumption, the financial system prefers the risky project, given the government guarantees a riskless payoff to the financier. In our model this means that, even though the government does not guarantee any payoff directly to the bank, it still incentivizes the financial system to take risk through a guaranteed payment to the financier in the crisis state, leading to lower capital costs for the bank if it chooses to take risk. In this setting, moral hazard between the financier and the bank does not exist, since a new Nash equilibrium can be achieved, with the guarantees of the government. However, this imposes a new problem of moral hazard between the financial system and the government, i.e. taxpayers. The government cannot control the actual risk taken by the financial system, even though it will bail out the system in case of a systemic crisis.

3.2 Impact of government guarantees on banks' MBV

We now would like to find out how these findings connect to the banks' market-to-book value of equity. Since the bank contributes N in the first state of our two-state-model, this reflects

the book value of equity. We present a way to describe the market value of equity. Again, we start assuming there are no government guarantees.

3.2.1 MBV without government guarantees

We show in (section 3.1.1) that in the absence of government guarantees, the bank would choose the riskless project, if no moral hazard exists. Nevertheless, we show that the market-to-book value would be the same, no matter if both participants could agree on the riskless project or not:

• When choosing the riskless project, the return for the bank would be R^f , and the payable amount to the financier would be $R^F = \rho \times R^f$, leading to a present value of the market value of equity in the first state of:

(10)
$$M = \frac{1}{R^f} \times \left[(R^f - \rho \times R^f) \times I - N + M \right] = N$$

• When choosing the risky project, the return for the bank would be R^n in the normal state and zero in the crisis state, the financier would require a return equal to $R^F = \frac{\rho \times R^f - q_c \times R^c}{(1-q_c)}$. The present value of the market value of equity in the first state of:



Using the fact that $(1 - q_c) \times R^n + q_c \times R^c = R^f$ and $\frac{N}{1 - \rho} = I$ we can rewrite this term as:

(12)
$$M = \frac{1}{R^f} \times (1 - q_c) \times \left[\frac{1 - \rho}{1 - q_c} \times R^f \times \frac{N}{1 - \rho} - N + M\right] = N$$

Hence, no matter which project the bank chooses (this depends on our assumptions), without government guarantees, M = N and, hence, MBV = 1.

3.2.2 MBV with government guarantees

If we extend the model again and include government guarantees, we rewrite the definition of *M* above, but set $R^F = \rho \times R^f$. We get

(13)
$$M = \frac{1}{R^f} \times (1 - q_c) \times \left[\left(R^n - \boldsymbol{\rho} \times \boldsymbol{R}^f \right) \times I - N + M \right] > N$$

We know that $\rho \times R^f > \frac{\rho \times R^f - q_c \times R^c}{(1-q_c)}$, the rest of the formula is equal to (11). We can see that in this case, the market value of equity is larger than in (11), hence M > N, leading to a MBV larger from one. An illustration of this using a simple example can be found in the appendix (example 1).

Looking at this model, anticipated government guarantees lead to debtholders taking their contribution for granted by the state and, hence, for risk-free, regardless of the bank's exposure to systemic risk. This leads to a higher expected return of risky projects and, hence, to higher cash flows in normal times. These higher future cash flows are priced in the market value of equity, leading to an MBV greater than one. However, this MBV greater than one also reflects higher losses of banks in case of a crisis, given government guarantees are considered in the market value of equity.

3.3 Comparative statics

We have seen that, given the model holds, a MBV higher than one is an indicator for anticipated government guarantees influencing decision making in the financial market. Now, we want to see if we can also infer something about the size of the MBV. Does a relatively high MBV indicate a relatively high tail risk and vice versa?

First, we determine how a change in R^N would influence the MBV, by computing the sensitivity using the partial derivative of equation (13).

We compute the MBV given government guarantees from (13):

$$\frac{M}{N} = \frac{1}{N} * \frac{s}{1-s} * [(R^N - \rho * R^f) * I - N]$$

With $s = \frac{1-q_c}{R^f}$

And derive it with respect to the return in normal times.

$$\frac{\partial M/N}{\partial R^N} = \frac{I}{N} * \left(\frac{s}{1-s}\right)$$

The partial derivative after return in normal times is positive if all two parts are positive:

- $\frac{1}{N}$: This term is the asset-to-equity ratio, which is a form of leverage. The higher the asset-to-equity ratio, the higher the leverage and the higher the effect of a higher tail risk on MBV. Since assets and equity are never negative, this term will always be larger or equal to one.
- $\frac{s}{1-s}$: This term comes from discounting for uncertainty and opportunity cost. Since s will be larger than zero, but smaller than one, the term will be larger than zero as well.

Hence, the MBV reacts positively to an increase in return in the normal state. The multiplier depends on leverage and the probability of the crisis and the discount rate. From (1), we know that an increase in return in the normal state must lead to a decrease in the return in the crisis state. Therefore, we compute how a change in R^c would influence the MBV, by plugging in equation (1):

$$\frac{\partial M/N}{\partial R^c} = -\left[\frac{I}{N} * \left(\frac{s}{1-s}\right) * \frac{q_c}{1-q_c}\right]$$

We see that this term is similar to the derivation above, with a negative sign and a third term:

• $\frac{q_c}{1-q_c}$: This term describes the ratio of the risk-neutral probability of the crisis state and the probability of the normal state, which is positive. The smaller the probability of the crisis state, the smaller the term.

The model thus predicts, that a change in the riskiness of the return in the crisis state will lead to an adverse change in the market-to-book value. Hence, a decrease in the crisis return, or an increase in tail risk, will lead to an increase in the MBV, given anticipated government guarantees are considered. Furthermore, leverage seems to have an impact on correlation between crisis return and MBV as well. The model indicates that a higher level of leverage leads to a stronger correlation of tail risk and MBV.

3.3.1 Limitations

The model described above explains the connection between the variation in tail risk (R_c) and the variation in MBV of banks. In section 5, we will test this connection with an empirical model. We take a closer look at the recent financial crisis and regress the returns of banks during the crisis ("tail risk") on their MBV in 2006 ("normal times"). Having this empirical study in mind, it is important to elaborate on the limitations of the theoretical model described before.

First, the model applied is a highly simplified model, which only focuses on the presence of anticipated government guarantees affecting tail risk and, hence, the MBV of banks. In reality, there is a number of other factors such as investors' beliefs or other factors not captured in the book value of equity (see section 3.4.1). However, such other factors are difficult model and might depend on factors attributable to investors' discretion.

Second, we use a two-stage model to describe continuous events. In the real world, we cannot easily differentiate between the normal and crisis state. In our empirical model we proxy the normal state, thus the state of the economy, which is unaffected by crises, through taking market and accounting data from the pre-crisis year (2006). After, we aggregate the average returns from the three years during the financial crisis to compute our proxy for returns in crisis times. In reality, tail risk (as the *possible* returns in case of a crisis) directly affects MBV without any time lag.

Third, it is very tough to empirically test causality. Our theoretical model shows the mechanism that the differentiation between normal and crisis state returns in the two-stage model leads to an MBV greater than one in the presence of government guarantees. We further show that a decrease in the crisis return leads to an increase in market-to-book value. In our empirical model we regress tail risk on MBV, which allows us to formulate conclusions on correlation, not on causality.

Fourth, since our model is a representative agent model with only one bank and one financier, it only explains a partial view. In our empirical model, we look at over 400 individual banks, which differ in their characteristics regarding asset portfolio, leverage, profitability, etc. However, even with these differences, we use the theoretical model to understand the mechanism behind the regression.

3.4 Hypotheses development

We have seen that the MBV is correlated to a bank's tail risk in the presence of government guarantees. In this section, we further describe what these findings mean for the economy and we present some literature in this area. Furthermore, we formulate three main hypotheses that we expect to approve in our empirical study in section 5, where we regress banks' tail risk on several measures including MBV, ROE and PE-ratio.

3.4.1 Market-to-book value

The MBV is defined as the market value of equity divided by the book value of equity of a company or the dollar price investors are willing to pay for every dollar of book equity of a firm. In literature the market-to-book value has been interpreted as a proxy for expected return on equity (Graham et al, 1962), as an indicator of mispricing of stocks (Rosenberg et al 1985) and as a proxy for risk (Fama & French, 1992). Conventional interpretations, such as the classification into value vs. growth stocks (Chan et al 1995), however do not make sense in the banking sector. The MBV for banks can have a different meaning compared to other industries, since banks usually have different types of assets. They only hold a few real assets, and most of the banking assets are often liquid and traded in the market or are very similar to traded assets. Hence, they are reported marked-to-market rather than on amortized costs, which is why banks' MBV should be closer to one. However, it will rise above one, if investors expect certain cash flows, which are not priced in the book value of equity (Bogdanova et al 2018). This is also referred to as franchise value or charter value (Keeley, 1990; Calomiris & Nissim, 2014; Sarin & Summers, 2016; Chousakos & Gorton, 2017). This difference between market and book value of equity, which can stem from beneficial market circumstances (Bogdanova et al 2018) or intangible assets entailing increased future profitability (Calomiris & Nissim, 2014) for example through a potential source of private customer information or through market entry barriers and oligopolistic behavior (Chousakos & Gorton, 2017). Another interpretation of the difference between market and book value of equity, however, can be due to implicit government guarantees (Atkeson et al, 2018), as described above.

To discuss implications of these theories, we first look at historical values. Figure 4 (see appendix) shows the historical development of the MBV of the US banking sector from 1976 to 2016, calculated from the data in our sample (see section 4.2). The value reflects the sum of the market value of equity of all banks divided by the sum of the book value of equity of all banks at year end. We observe a steady increase of the ratio until 1997, followed by a slight drop in MBV. This might be explained by the dotcom bubble. After, MBV rises again until 2006, the year before the global financial crisis, to over 3. The global financial crisis leads to a strong drop in banks' MBV to almost 1, followed by a slight recovery to over 2 in 2016. Since we do not observe a return to pre-crisis market-to-book values of banks in the years after the financial crisis (Sarin & Summers, 2016), this could entail two potential implications. First, it could mean that investor confidence in banks has not been fully restored, leading to lower equity valuation, potentially accelerated through stricter regulations. This would imply that

stricter regulatory measures hampered banks recovery leading to "banking illness" (Chousakos & Gorton, 2017) and even increased market leverage (Sarin & Summers, 2016). On the other hand, according to Atkeson et al (2018) a lower post-crisis MBV valuation could simply mean that banks' pre-crisis MBV contained high valuations of potential government bailouts, which are less prevalent in post-crisis years due to stricter regulations aiming to enforce lower risk-taking.

We follow this argument. The theoretical framework argues that the potential bailout through taxpayers increase the safety of banks' debt and thus decreases the payable interest rate, which means that risky projects become more profitable for banks. This leads to banks taking more risky projects, which is captured in the market value of equity. Hence, we argue that the difference between market and book value of equity stems from a rational source of pricing externalities, such as implicit government guarantees. Following, the MBV should increase with lower tail return (i.e. higher tail risk), given the presence of anticipated government guarantees. If this assumption is true, we should find a negative relationship between MBV and tail return. This leads us to our first hypothesis.

H1: There is a negative relationship between the market-to-book value in normal times and returns on bad days.

If this hypothesis holds, it is a strong sign for the presence of government guarantees. Moreover, our model suggests that the effect should be stronger with higher leverage.

3.4.2 Return on equity and return on assets

Both ROE and ROA are commonly used profitability ratios to compare the financial performance across firms. ROE is defined as the earnings attributable to equityholders, usually net income, scaled over the book value of equity. ROA is defined as the earnings attributable to equity- and debtholders, usually earnings before interest expense (EBIE), scaled over the book value of assets.

In the two-state frictionless model described above, we assume that the investor, after applying risk-neutral probabilities, will expect the same return from the risky project than the risk-free rate. This means that under the assumption of a constant risk-free rate, an increase in the expected return on assets in the normal state will have an adverse effect on the expected return on assets in the crisis state and vice versa. ROE, hence, should increase with tail risk, i.e. decrease with the return on crisis days.

Looking at risk and ROE, we always have to take into account leverage as well. Since investment decisions are made by equityholders and investors, banks are focused on increasing ROE instead of ROA, which might create false incentives. ROE increases with leverage ceteris paribus in a setting with discriminating taxes, which gives incentives to increase leverage (Modigliani & Miller, 1958). This effect is even stronger in the presence of government guarantees, which leads to extremely cheap leverage. Furthermore, low interest rates on debt leads to banks underestimate their own asset risk. This leads to increased leverage ratios and higher risk, reflected in a higher return on equity (Acharya & Franks, 2008).

Meiselman et al (2018) directly investigate the correlation of ROE and tail risk, using the same method that we are using in this paper. They find a significant relationship between the return on equity of American banks in 2006 and their stock market returns on the worst days of the financial crisis (as a proxy for tail risk). They argue that banks can be looked at as asset portfolios with a certain exposure to systemic risk. The higher this exposure, the higher the returns, following the CAPM.

The historical development of the average ROE (see appendix, figure 5) shows a constant varying ROE of around 0.14 from 1976 to 2006, when we observe a strong decrease in banks' ROE during the global financial crisis to almost 0. Again, we observe a slight recovery of ROE after 2010 to around 0.11 in 2016. In line with our argumentation for MBV, this difference between pre- and post-crisis levels of ROE could either indicate a hampering of banks' business models or lower risk taken (Meiselman et al, 2018).

We, hence, also expect a negative relationship between ROE and return on crisis days in our regressions.

3.4.3 Price-earnings ratio

The price-earnings ratio is defined as price per share over earnings per share or market capitalization divided by earnings. As a valuation multiple, it reflects the price investors are willing to pay per unit of earnings. Since the market value of equity can be reformulated as expected future earnings, the PE ratio can be interpreted as the ratio between future vs. current earnings of a firm. A relatively high PE-ratio, hence, means that investors are willing to pay more for a certain company compared to another company with the same current earnings in expectation of stronger earnings growth. Interpretation of the PE-ratio range from being an earnings growth indicator (Cragg & Malkiel, 1982) to a risk factor (Ball, 1978). From looking at the historical development, the average PE has steadily increased in our time frame from

1976 to 2016 (see appendix, figure 6). Interestingly, the main trend, after accounting for drops of the ratio after the dotcom bubble or the global financial crisis, is still increasing, not showing an indication of mean reversal.

Existing literature about the price-earnings ratio presents contradictory findings. On the one hand, high PE ratios are found to be followed by high returns, leading to mean reversion of the ratios in the long-term (Campbell & Shiller, 1998), indicating that market expectations are correct on average. On the other hand, persistently increasing PE, have often initiated crises and bubble (Collyns & Senhadji, 2002), which are a source of irrational market expectations. Hence, the effect of PE is two-edged: If investors were rationally expecting a crisis, low PE ratios should precede firms with a high tail risk, however if investors had irrational beliefs, a high PE ratio should be an indicator of a high tail risk.

3.4.4 Decomposition of MBV into ROE and PE-ratio

We have seen that MBV and ROE should, in theory, have a negative correlation with banks' returns in crisis times. Furthermore, it is not clear how PE-ratios and tail risk are connected to each other.

Since market-to-book value can be decomposed into return on equity and price-earnings ratio, and return on equity is not found to indicate the price-earnings ratio (Penman,1996), we explore whether the effect of the MBV on tail risk can be broken down into an effect from ROE and an effect from PE.

$MBV = ROE \times PE$

This shows us that all the information contained in ROE and PE is also included in the MBV. The MBV, hence, should always explain at least as much as ROE and PE together. This leads us to our second hypothesis:

- *H2a: Return on equity does not explain any variation in tail risk that cannot be explained with the market-to-book value.*
- *H2b: Price-earnings-ratio does not explain any variation in tail risk that cannot be explained with the market-to-book value.*

By taking logarithms, we can decompose the equation into an addition:

$$\ln(MBV) = \ln(ROE) + \ln(PE)$$

Hence, we propose a logarithmic relation between ROE, PE and MBV. This should lead to the following regression:

$\ln(return_{crisis}) = \beta \times \ln(\text{MBV}) \equiv \beta \times \ln(ROE \times PE) \equiv \beta \times \ln(ROE) + \beta \times \ln(PE)$

This means that, by regressing the logarithm of tail risk on the logarithm of PE and ROE, we should get the same coefficients for PE and ROE. If this is the case, we can conclude that ROE and PE together do not contain any information that is not already contained in the market-to-book value, which is our third hypothesis:

H3: Return on equity and price-earnings ratio together do not contain any information that is not contained in the market-to-book value.

In the next section, we will describe our methodology to investigate the hypotheses developed above in detail, before we provide our findings in section 5.

4. Research design and methodology

The goal of this thesis is to find out more about the relationship between banks' tail risk and their market-to-book value. In the last section, we have described the theories behind this issue. Now, we would like to test our findings through several regressions. The main regression of interest will be:

$$return_i^{crisis} = \beta \times MBV_i^{pre-crisis}$$

In this section, we further describe how we design our empirical study and which data we use.

4.1 Research design

Using empirical data from 470 US banks in the time shortly before and during the recent financial crisis, we would like to test if our empirical results confirm the mechanisms from the theoretical model. We investigate whether there should be a positive relation between the MBV of a bank and its tail risk (i.e. a negative relation between MBV and crisis return), since this gives us a hint about whether government guarantees priced in the MBV serve as a risk factor, because they incentivize the financial system to take risk. Then, we should find evidence that possible future cash flows from governments payouts were reflected in banks' market values of equity in 2006, the year prior to the financial crisis. It is important to mention that we look at the tail risk of the individual banks, since it is a proxy for their exposure to systemic risk.

We will investigate whether we find such a connection using the methodology developed by Acharya et al (2016b) and used by Meiselman et al (2018). As described in section 2, it is not possible to directly observe tail risk. We will, hence, proxy this risk for our regression through the marginal expected shortfall (MES) (see section 2). Since we can only measure the MES ex post, we have to work with past data. To test whether MBV in normal times is a good measure for tail risk, we regress the MBV of banks in 2006, the year before the recent financial crisis, on their MES during the crisis. MES during the crisis is, hence, a proxy for tail risks banks were exposed to in 2006, just before the crisis, i.e. in normal times.

In line with Meiselman et al (2018), in order to calculate MES, we use stock market returns since holding period returns act quickly on recent events.

4.2 Data and methodology

The sample is constructed in a similar manner to Meiselman et al (2018). We examine daily stock market returns, annual income and balance sheet statements for financial firms in the U.S. Our source for daily stock market data is from the database of the Center of Research in Security Prices (CRSP), accessed through Wharton Research Data Services (WRDS), and we access the merged CRSP/Compustat database for data on annual income and balance sheet data.

Daily stock market data is taken in the time period of 1 July 1926 - 31 December 2016 in the stock markets NYSE, AMEX, NASDAQ & ARCA. We limit our stock data to firms with available data and with actively traded, ordinary common shares.

Accounting data is taken in the time period of January 1976 - December 2016 per fiscal year. Financial firms are defined as banks included in the CRSP/Compustat joined database bank annual, which includes banks with the following Standard Industrial Classification (SIC) codes: 6020, 6021, 6022, 6029, 6035 and 6036. This includes different types of commercial banks and savings institutions.

We aggregate daily stock market data and accounting data on the basis of PERMCO and create an unbalanced panel of 470 observations. We use the following ratios for our main regression, detailed definitions can be found in appendix, table 17:

$$ROE = \frac{Net \ income}{Tangible \ book \ equity} \qquad ROA = \frac{EBIE}{Book \ Assets}$$
$$MBV = \frac{Market \ value \ of \ equity}{Tangible \ book \ equity} \qquad PE \ ratio = \frac{Market \ value \ of \ equity}{Net \ income}$$

For profitability measures, we use the return on assets and return on equity. We construct our return on assets correspondingly to Meiselman et al (2018) by dividing the earnings before interest expenses (EBIE) by book assets. We use EBIE as a gross profit measure to proxy earnings to equity- and debtholders, calculated by adding back interest expenses and taxes to net income. Alternatively, we add interest expense back to pre-tax income, in case the data is not available.

For the return on equity, we calculate the return on tangible equity, by dividing net income by tangible equity. We choose the measure of tangible equity since recognition of intangible assets in the balance sheet is loosely defined and thus can be subject to the discretion of company

accounting (IAS 38). Also, in a case of liquidation intangible assets might not be liquid enough to sell it for the recognized value, which differentiates it from tangible assets. Hence, we exclude intangible assets to have a clean measure, which is less subject to creative accounting, in line with Meiselman et al (2018).

We construct the market-to-book value as market value of equity, obtained through the daily stock data, divided by tangible book equity. We define the price-earnings ratio as the market value of equity over net income. Since we take both measures from the end of the year, this equals the common measure of share price over earnings per share (for details, see appendix, table 17).

Following Meiselman et al (2018) and Acharya et al (2016b) we construct a measure for banks' tail risk by taking average holding period returns during the worst days in the financial crisis. This is done in the following way: First, we define the worst 5% days from 1 July 1926 - 31 December 2014. After, we take the average holding period returns of banks on the subset of the 5% worst days that occurred in the financial crisis.

We define "5% worst days", by constructing a measure for the worst days in the financial sector using the 5% lowest value weighted returns from the Fama and French 48 Industry Portfolio (industry 44 of 48), thus "bad bank days". Meiselman et al (2018) compares these results with the alternative measure of "bad market days" as the worst days of the market, by identifying the 5% worst value weighted returns of the entire market portfolio. Even though both measures have different interpretations, we find similar results for the main regression. In the following parts, we will only use "bad bank days" to approach the worst days for computing tail risk.

Following, we compute the simple annual average of daily stock market returns of these worst days during the financial crisis for each company on the basis of the permanent company identifier (PERMCO), which is our measure of systemic risk, denoted as tail risk, per bank.

We standardize all our variables to have a mean of zero and a standard deviation of one and run several regressions in the following style:

$$return_i^{crisis} = \beta_0 + \sum \beta_j * variable_i^{pre-crisis}$$

5. Findings

We explore the relationship between our market and accounting values on the tail risk through several regressions. In table 2, we compare the regression using MBV to the regression using return on equity, return on assets and price-earnings ratio. We show that MBV is a better explanatory variable in our regression on tail risk than ROA, ROE and PE by looking at significance, the coefficient and the partial R². In our second step, we expand on the question whether ROE and PE individually have a higher explanatory power than the MBV, applying a logarithmic decomposition (table 3). By applying the logarithmic decomposition and finding the estimates for the logarithmic values we find that ROE and PE together do not significantly explain more than only looking at MBV. Finally, we explore the relationship between the market-to-book value and other ratios, revealing about an individual bank's characteristics. In an application, we try to predict the tail risk in 2016 with a model using different variables and compare this to the tail risk estimated for and in 2006. Using two models, we do not find a significant difference in 2006 and 2016 levels of tail risk.

5.1 Main regressions

For our main regressions, we obtain the following summary statistics in table 1. The return on assets varies from 2.02% to 7.93%, with a mean of 4.1%. The ROE varies from 1.13% to 78.68%, with a mean of 13.7%. The MBV value varies from 0.81 to 8.84 with a mean of 2.39 and the price-earnings ratio varies from 4.84 to 129.5 with a mean of 21.68.

Table 1: In this table, we show the summary statistics of key variables used for the regressions. "Bad bank days" are the 5% worst performing days of the Fama French banking industry portfolio from Jul 1926 to Dec 2014 with the lowest-value weighted return. "Bad market days" are the 5% worst performing days of the value weighted market index return from Jan 1926 to Dec 2014 with the lowest-value weighted return. Average bad-day returns are calculated from bad days between 1 Sep 2007 and 1 Oct 2010. The construction of the variables is explained in table 17.

Statistic	Ν	Mean	St. Dev.	Min	Max
Return on bad bank days (%)	470	-2.10	1.68	-6.98	1.52
Return on bad market days (%)	469	-2.50	1.95	-7.91	3.31
Equity (billions)	470	0.78	4.07	0.01	55.75
Assets (billions)	470	16.07	102.45	0.10	1,459.74
Return on Assets (%)	470	4.10	0.77	2.02	7.93
Return on Equity (%)	470	13.70	7.96	1.13	78.68
Market-to-book value	470	2.39	0.97	0.81	8.84
Price-earnings ratio	470	21.68	14.81	4.84	129.50

Summary statistics for main regression

Table 2 shows the regression results for the estimation of tail risk using MBV. In this table we proxy tail risk by looking at the average return on the "bad bank days". By looking at "bad market days", we obtain similar findings (see appendix table 12). In table 2, we see a negative significant correlation between the MBV and return on bad days (1). An increase in the MBV of one standard deviation leads to a decrease of 0.540 standard deviations in the return on bad bank days. We compare these to the variables of the original Meiselman et al (2018) paper, ROA (2) and ROE (5) in terms of estimate, significance and coefficient of determination. Due to standardization we can compare the estimates of the different regressions. We see that the estimate of the MBV (-0.540) is stronger negative than the estimates of both ROA (-0.280) and ROE (-0.441) and, hence, is a larger factor in explaining tail risk variations. We observe a higher t-value, hence a higher significance, of MBV (-13.88) than both ROA (-6.31) and ROE (-10.62). However, all three variables are significant at the 0.1%-level in the simple regression. PE shows a positive significant estimate (0.095), in contrast to the other three variables, with a lower significance level (5%). Hence, an increase in PE leads to a decrease in tail risk, i.e. a higher crisis return. The coefficient of determination, R^2 , is a measure of goodness of the "insample" fit of the dependent variable. MBV (0.292) shows a better "in-sample" fit than ROA (0.078), ROE (0.194) and PE (0.009).

Table 2: This table presents results from a set of OLS regressions that estimate the relationship between tail risk and prior accounting and market ratios with the regression model:

$$return_{i}^{crisis} = \beta_{0} + \sum_{i} \beta_{j} * variable_{i}^{pre-crisis}$$

		Kegr	ession Res	suits. Dau	Dalik uay	3				
	Dependent variable: Return on bad bank days/ tail risk									
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)		
MBV	-0.540***		-0.501***		-0.502***		-0.536***	-0.521***		
	(-13.876)		(-12.574)		(-8.048)		(-13.686)	(-7.175)		
	[0.292]		[0.253]		[0.122]		[0.286]	[0.099]		
ROA		-0.280***	-0.147***							
		(-6.307)	(-3.681)							
		[0.078]	[0.028]							
ROE				-0.441***	-0.048			-0.020		
				(-10.622)	(-0.767)			(-0.241)		
				[0.194]	[0.001]			[0.000]		
PE						0.095^{*}	0.035	0.026		
						(2.057)	(0.888)	(0.507)		
						[0.009]	[0.002]	[0.001]		
Intercept	-0.000	0.000	0.000	-0.000	-0.000	0.000	-0.000	-0.000		
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)		
Observations	470	470	470	470	470	470	470	470		
\mathbb{R}^2	0.291	0.078	0.311	0.194	0.292	0.009	0.293	0.293		
Adjusted R ²	0.290	0.076	0.309	0.193	0.289	0.007	0.290	0.288		
Note:					p-values: *	p<0.05 **	p<0.01 ***	p<0.001		

All variables are standardized to have a mean of zero and a standard deviation of one.

t-values in (), partial r^2 in []

In the second step, we look into the multivariate regression with both MBV and ROA/ROE/PE, number (3), (5), (7). Adding the corresponding return variable to the regression with the market-to-book value leads to slight increases in R^2 and adjusted R^2 . However, MBV still seems to be the variable that explains more of the dependent variable, since it has a much higher partial R^2 , which gives us the percentage of explanation that is explained by one but not the other variable. The high value of partial R^2 for the MBV (0.253) in (3), in contrast to the much lower value for the ROA (0.028), indicates that the explanatory power of the MBV is much stronger compared to the return variable. The same applies for regression (5) with ROE (0.122 for MBV, 0.001 for ROE) and (7) with PE (0.286 for MBV, 0.002 for PE). We also note that the estimates and t-stats remain high for the MBV, while being much lower for the ROA, ROE and PE.

If we regress tail return on MBV, ROE and PE together (number (8)), we get a similar picture. The high significance and partial R^2 of MBV indicates that the explanatory power stems mainly from that value, while both ROE and PE are insignificant.

The high significance of regression (1) leads us to the conclusion that H1 (see section 3) is true. MBV seems to explain a significant part of the variation in tail risk. Looking at (4) and (6) also supports what we have expected: Both, ROE and PE have a significant correlation with tail risk. However, both ratios lose their significance if we add MBV to the regression ((5), (7) and (8)). This leads us to also approve H2a and H2b: ROE and PE do not explain anything that is not already explained in the MBV.

For the logarithmic transformation, we take the natural logarithm of the independent variables above. For the tail risk variable, we exclude any average crisis return which is positive, and take the absolute value before applying logarithms. Although we have 440 numbers of observations after this restriction, this allows us to not perform other transformations, which might skew the results (see table 13 for summary statistics). We do not standardize these variables, since the interpretation would be difficult with standardization and logarithmic transformation.

Table 3: This table presents results from a set of OLS regressions that estimate the relationship between tail risk and prior accounting and market ratios with the regression model:

$$\ln(return_i^{crisis}) = \beta_0 + \sum \beta_j * \ln(ratiable_i^{pre-crisis})$$

For the logarithmic decomposition, we ignore positive numbers for return on bad bank days in order to apply logarithms.

			Dependen	t variable:					
-	log return on bad bank days/ tail risk								
	(1)	(2)	(3)	(4)	(5)	(6)			
log MBV	1.263***		1.238***		1.261***				
	(11.742)		(8.426)		(11.639)				
	[0.239]		[0.14]		[0.237]				
log ROE		0.530^{***}	0.023			1.261***			
		(7.581)	(0.256)			(11.639)			
		[0.116]	[0.000]			[0.237]			
log PE				-0.131	-0.023	1.238***			
				(-1.297)	(-0.256)	(8.426)			
				[0.004]	[0.000]	[0.14]			
Intercept	-5.167***	-3.004***	-5.098***	-3.749***	-5.098***	-5.098***			
	(-53.53)	(-19.341)	(-17.74)	(-12.473)	(-17.741)	(-17.741)			
Observations	440	440	440	440	440	440			
\mathbb{R}^2	0.239	0.116	0.240	0.004	0.240	0.240			
Adjusted R ²	0.238	0.114	0.236	0.002	0.236	0.236			
Note:			р	-values: *p<	0.05 **p<0.01	l ****p<0.001			

MV BV Decomposition

t-values in (), partial r² in []

Looking at table 3, we see, for (1), (2) and (4), a change in sign of the estimate and the t-statistic compared to table 2. Since we take absolute values of tail returns before taking logarithms, the previous negative estimates of the MBV and ROE are now positive. The PE ratio shows a negative estimate, since it was positive. Due to these transformations, we can interpret a 1% increase in the market-to-book value as a resulting 1.3% decrease in the return on bad bank days.

We again see that ROE (1% level) and PE (10% level) are significant alone ((2) and (4)) but lose their significance if we add MBV to the regression ((3) and (5)). We obtain the same low absolute t-stat (0.256 vs. -0.256), reflecting a p-value of 79.8%. Due to the decomposition (section 3.4.4) we know that if the coefficient of MBV in (1) is equal to the coefficients of ROE

and PE in (6), ROE and PE together do not explain any variation in tail risk that is not explained by MBV. Hence, we compute the F-value for the joined significance of ROE and PE, by testing the probability for the coefficients of the unrestricted model (6) to equal the coefficient of the restricted model in (1). We again obtain a p-value of 79.8% (F-value = 0.066, df₁ = 437, df₂ = 1), emphasizing that the ROE and PE together do not significantly explain more than the MBV alone. We conclude that the explanatory power from both ROE and PE is included in the MBV. Hence, we cannot reject H3.

Overall, our regressions show what we expected from our theoretical model: Both, ROE and PE explain some of the variation of tail risk. However, together they do not explain more than the MBV alone, giving an indication that the explaining factor might stem from the MBV. In the simple regression, MBV explains around 29% in the variation of tail risk. This means that MBV can, at least, partly serve as a proxy for banks' exposure to systemic risk. Our theoretical model implies a possible mechanism: market participants might have priced possible government bailouts in the share price of banks before the recent financial crisis.

5.2 Robustness tests

We have seen that MBV can explain tail risk and has more explanatory power than ROE and PE. Our robustness test explores if the findings are robust under slightly different assumptions. First, we add control variables, which might be correlated with an institutions' tail risk. Thus, we also explore if the explanatory power of the MBV stems from correlation with another factor. We choose the common accounting ratios and try to link them qualitatively to our theoretical explanation. In specific, we test for the control variables size, book leverage, market leverage, non-interest income, interest income, intangibles, loans, deposits and dividends (see appendix, table 17 for definition of the control variables). Second, we explore the relationship between MBV on tail risk for varying levels of leverage, since our theoretical model predicts an increase in the sensitivity of MBV for higher levered firms. Finally, we re-estimate the main regression using quantile-regression to test whether the results are applicable to other quantiles.

5.2.1 Control variables

Size. We are controlling for size since we expect a negative relationship between size and tail risk. The larger the firm, the more important should it be for the whole financial system. In case of a significant institution, the bad days in the economy could even stem from a drop in the return from that firm, reflected directly in our measure for firm-specific tail risk.

Additionally, large institutions are often "systemically important institutions" and due to their size more interconnected with other banks, which are both reasons why these large firms would expect government guarantees in case of a breakdown. Literature finds that size is a primary factor for explaining systemic risk (Varotto & Zhao, 2018; Laeven et al 2016). Vallascas & Keasey (2012) find that a size cap for firms would decrease systemic risk. Hence, we expect size to be another factor explaining tail risk, since a larger firm is more important for the economy and also more likely to expect government guarantees.

Leverage. Leverage is an important control variable, since a higher leverage changes the risk preference of equityholders due to gambling for resurrection (see also section 2.2.2). Firms with higher leverage would be more likely to take on risk and take higher advantage of government guarantees. We control for both book and market leverage. Book leverage is closely related to regulators measures of capital requirements (i.e. CET 1). However, market leverage includes the market value of equity, which provides us with an indicator of how market equity is valued. The theoretical model shows that the relation of tail risk and MBV is stronger for higher levered firms. Acharya & Franks (2008) show in their model, that low interest rates, due to government guarantees, lead to incentives for firms to have higher leverage. Acharya & Thakor (2016) indicate in their model, that higher leverage would induce banks to choose liquidation, and other literature document the relationship between leverage and banks' balance sheet decision (Adrian & Shin, 2010).

Non-interest and interest income. Non-interest income and interest income are indicators of the banking activities the firm engages in. Whereas interest income is attributed to the traditional deposit taking and lending business of the firm, non-interest income is associated with investment banking, asset management, trading and securitization businesses (Brunnermeier et al 2012). Previous studies have found that a higher share of non-interest income is correlated with a higher systemic risk (Brunnermeier et al 2012) and the predictiveness of the return measure on tail risk is stronger for non-interest income (Meiselman et al, 2018). Meiselman et al (2018) offer a possible explanation by linking the non-interest income to high-risk banking activities outside of the core lending business.

Intangible assets. We further account for intangible assets. The concept of intangible assets is closely related to franchise value, such as firm brand, customer rights or intellectual property leading to future growth opportunities (Chousakos & Gorton, 2017). However, due to accounting standards, not all types of intangible assets are allowed to be recognized on the balance sheet, only if they are purchased or internally developed according to IAS (IAS 38).

We use these book values of intangible assets, which is why we can only proxy for franchise value. Hence, including this control variable allows us to partly separate the effect of government guarantees from the one of franchise value.

Loans and deposits. We add a bank's share of loans and deposits as control variables. Loans alone do not give us an indication about the riskiness of the banks' portfolio, since the latter also depends on other activities, i.e. engagement in trading activities. Similarly, deposits also do not give us an indication of the stability of financing side of banks, since they depend on banks' financing structure. Adrian & Brunnermeier (2016) differentiate between non-interest bearing deposits, which are allocated across banks in case of stress, and interest-bearing deposit, which are more stable sources of funding. In their estimation of CoVaR, non-interest bearing deposits positively and interest-bearing deposits negatively predict future tail risk. However, we do not differentiate between these types of deposit in this set of control variables.

Dividends. We include dividends as a control variable, since Acharya et al (2017) argue that dividend payout policy can exert negative externalities on other banks, thus increasing systemic risk. Meiselman et al (2018) find an empirical relationship between higher dividends and larger tail risk. First, firms with high earnings on equity would issue dividends, since dividend persistence is expected from investors. Second, dividends have a strong signaling power, hence strong future returns on shares are expected. Both of these arguments strengthen our existing relationship between return and tail risk and might be reasons for taking higher systemic risk and taking advantage from government guarantees.

Beta. As another control variable, we add the one-year CAPM-beta (Sharpe, 1964). The measure is different to all other measures used in this thesis. Since we want to find out whether market and book values of banks can give us information about their tail risk, we do not work with modelled and estimated values. Beta, however, is a proxy for co-movement with the market that cannot be directly observed and has to be estimated. We calculate the firms' covariance with the market divided by the variance of the market in the pre-crisis year, as a modelled measure for the firm's co-movement with market returns. We add beta as control variable since it has been showed that the measure can actually explain firms' risk, even though it does not explicitly focus on tail risk. However, using systematic risk to estimate tail risk might not add further explanatory value, since both are measures driven by the performance of the banking sector or market, respectively. If a bank would choose risky projects to increase its earnings and MBV, this would likely lead to higher tail risk and higher systematic risk, measured in a higher beta.

Again, we run the following regression:

$$return_{i}^{crisis} = \beta_{0} + \beta_{1} * MBV_{i}^{pre-crisis} + \sum \beta_{j} * control_{i}^{pre-crisis}$$

We run the regression with our tail risk measure, MBV and each of the control variables separately (Table 4 and 5), in random combinations and all together (Table 6), again under standardization, so we can compare the estimates. We obtain the following results: In all separate control regressions, our estimate for market-to-book is still significant at the 5%- level, indicating that the findings are robust.

Table 4: This table presents results from a set of OLS regressions that estimate the relationship between tail risk and MBV and other control variables with the regression model:

$$return_{i}^{crisis} = \beta_{0} + \beta_{1} * MBV_{i}^{pre-crisis} + \sum_{i} \beta_{j} * control_{i}^{pre-crisis}$$

All variables are standardized to have a mean of zero and a standard deviation of one.

			11000100						
		De	ependent variab	ole:					
-	Return on bad bank days/ tail risk								
	(1)	(2)	(3)	(4)	(5)				
MBV	-0.177***	-0.565***	-0.599***	-0.555***	-0.558***				
	(-4.655)	(-14.931)	(-15.238)	(-13.041)	(-14.302)				
	[0.046]	[0.331]	[0.34]	[0.274]	[0.312]				
Size	-0.643***								
	(-16.961)								
	[0.389]								
Book leverage		-0.206***							
-		(-5.429)							
		[0.061]							
Market leverage			-0.174***						
_			(-4.433)						
			[0.042]						
Non-Interest				-0.009					
Income				(-0.206)					
				[0.000]					
Interest Income					0.021				
					(0.527)				
					[0.001]				
Intercept	0.000	0.000	0.000	0.000	0.000				
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)				
Observations	454	454	454	454	454				
\mathbb{R}^2	0.580	0.354	0.341	0.312	0.312				
Adjusted R ²	0.578	0.351	0.338	0.309	0.309				
Note:			p-values: ³	*p<0.05 **p<0.	01 ***p<0.001				

Regression Results

t-values in (), partial r² in []

Table 5: This table presents results from a set of OLS regressions that estimate the relationship between tail risk and MBV and other control variables with the regression model:

$$return_{i}^{crisis} = \beta_{0} + \beta_{1} * MBV_{i}^{pre-crisis} + \sum_{i} \beta_{j} * control_{i}^{pre-crisis}$$

All variables are standardized to have a mean of zero and a standard deviation of one.

			Results						
	Dependent variable: Return on bad bank days/ tail risk								
-									
	(1)	(2)	(3)	(4)	(5)				
MBV	-0.472***	-0.559***	-0.560***	-0.514***	-0.306***				
	(-10.918)	(-14.246)	(-14.929)	(-10.812)	(-8.611)				
	[0.209]	[0.310]	[0.331]	[0.206]	[0.141]				
Intangibles/Assets	-0.186***								
	(-4.301)								
	[0.039]								
Loans/Assets		-0.007							
		(-0.167)							
		[0.000]							
Deposits/Assets			0.229^{***}						
			(6.101)						
			[0.076]						
Dividends/Equity				-0.077					
				(-1.614)					
				[0.006]					
Beta					-0.549***				
					(-15.453)				
					[0.346]				
Intercept	0.000	0.000	0.000	0.000	0.000				
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)				
Observations	454	454	454	454	454				
\mathbb{R}^2	0.339	0.312	0.364	0.316	0.550				
Adjusted R ²	0.336	0.309	0.362	0.313	0.548				

Regression Results

Note:

p-values: *p<0.05 **p<0.01 ***p<0.001

t-values in (), partial R² in []

Table 6: This table presents results from a set of OLS regressions that estimate the relationshi between tail risk and MBV and other control variables with the regression model:

$$return_{i}^{crisis} = \beta_{0} + \beta_{1} * MBV_{i}^{pre-crisis} + \sum_{i} \beta_{j} * control_{i}^{pre-crisis}$$

All variables are standardized to have a mean of zero and a standard deviation of one.

			Regression	Results			
			Dep	endent varia	ble:		
			Return on	bad bank day	∕s∕ tail risk		
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
MBV	-0.129**	-0.556***	-0.499***	-0.560***	-0.073*	-0.571***	-0.112**
	(-3.011)	(-13.04)	(-11.859)	(-11.611)	(-1.971)	(-13.097)	(-2.419)
	[0.02]	[0.274]	[0.239]	[0.231]	[0.009]	[0.276]	[0.013]
Size	-0.666***				-0.519***		-0.532***
	(-15.812)				(-13.098)		(-13.096)
	[0.358]				[0.277]		[0.280]
Book leverage	-0.113			-0.203***			0.104
	(-1.701)			(-5.174)			(0.853)
	[0.006]			[0.056]			[0.002]
Market leverage	0.163*				0.012	-0.021	0.008
-	(2.390)				(0.388)	(-0.386)	(0.112)
	[0.013]				[0.000]	[0.000]	[0.000]
Non-Interest		-0.007		0.013		0.015	0.073^{*}
Income		(-0.166)		(0.317)		(0.362)	(2.382)
		[0.000]		[0.000]		[0.000]	[0.013]
Interest Income		0.020			-0.064*		-0.060*
		(0.512)			(-2.324)		(-1.775)
		[0.000]			[0.012]		[0.007]
Intangibles/			-0.146***				-0.051
Assets			(-3.428)				(-1.425)
			[0.026]				[0.005]
Loans/			-0.071				-0.085**
Assets			(-1.862)				(-2.635)
			[0.008]				[0.015]
Deposits/			0.220***			0.216***	0.161**
Assets			(5.701)			(4.093)	(2.012)
1 100000			[0.068]			[0.036]	[0.009]
Dividends/			[0.000]	-0.019		[0.050]	0.057
Equity				(-0.369)			(1.708)
Equity				[0.000]			[0.007]
Beta				[0.000]	-0 381***		-0.376***
Deta					(-11 883)		(-12 014)
					[0 240]		[0.25]
Intercent	0.000	0.000	0.000	0.000	0.000	0.000	-0.000
mercept	(0.000)	(0,000)	(0.000)	(0.000)	(0.000)	(0.000)	(0,000)
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Observations	454	454	454	454	454	454	454
\mathbb{R}^2	0.586	0.312	0.384	0.354	0.690	0.365	0.714
Adjusted R ²	0.582	0.308	0.379	0.350	0.687	0.359	0.707

Looking closer at the controls, we find significant estimates for size, book as well as market leverage, intangible assets, deposits and beta. The estimate for size has a negative sign and is highly significant, confirming our argumentation that larger firms tend to be more systemically relevant and experience greater losses on crisis days. Both book and market leverage show a negative coefficient. Hence, leverage indeed is a factor contributing to higher tail risk. We further find that intangible assets show a significant negative relationship with tail risk. Furthermore, we find a positive significant relationship between the share of deposits and tail risk, hence higher deposits lead to lower tail risk, i.e. higher returns in crisis times. We also observe a negative relationship between beta and tail risk. A higher beta is, therefore, a sign for greater losses in crisis times. This is interesting since it means that, even though focusing on general market co-movements, beta can still explain returns in extremely bad times.

We do not find consistent significant estimates for the remaining variables at the 5%-level. Looking at tables 4 and 5, which show regressions of MBV together with only one control variable, we see that its coefficient experiences the largest decrease in estimate and t-value when regressed together with size and beta. Hence, size and market co-movement of a firm seem to explain part of the variation of tail risk, which is explained by MBV if it serves as the only explanatory variable. However, in tables 4-6, we observe that the market-to-book value is still significant at the 5%-level and consistently shows a negative coefficient in all regressions, indicating that our findings from section 5.1 are robust.

5.2.2 Results under varying leverage

Since leverage is a much discussed factor in banking literature and our theoretical model also predicts a relation with the MBV, we further explore these two variables by dividing the sample into quartiles depending on their book leverage: 25% is the smallest quartile and 100% is the largest quartile. In table 7, we run the regression with MBV and book leverage.

Table 7: This table presents results from a set of OLS regressions that estimate the relationship of MBV and leverage on tail risk, after dividing the sample into quartiles depending on their leverage:

$$return_{i}^{crisis} = \beta_{0} + \beta_{1} * MBV_{i}^{pre-crisis} + \sum \beta_{2} * leverage_{i}^{pre-crisis}$$

All variables are standardized to have a mean of zero and a standard deviation of one.

	Dependent variable:							
-		Return on bad ba	nk days/ tail risk					
	25%-Quartile	50%-Quartile	75%-Quartile	100%-Quartile				
MBV	-0.508***	-0.691***	-0.584***	-0.519***				
	(-7.093)	(-8.713)	(-7.376)	(-7.002)				
	[0.312]	[0.408]	[0.331]	[0.306]				
Book leverage	-0.282	0.233	-0.607*	-0.144				
	(-1.166)	(0.616)	(-1.69)	(-1.178)				
	[0.012]	[0.003]	[0.025]	[0.012]				
Intercept	-0.059	0.091	0.181^*	-0.120				
	(-0.213)	(0.517)	(1.747)	(-0.638)				
Observations	114	113	113	114				
\mathbb{R}^2	0.316	0.409	0.337	0.308				
Adjusted R ²	0.303	0.399	0.325	0.295				

Regression Results: Varying leverage

Note:

p-values: *p<0.05 **p<0.01 ***p<0.001

t-values in (), partial r² in []

Since book leverage is not significant in these regressions except for the 75%-quartile, we cannot make profound statements about leverage changes within these quartiles. We observe a much stronger relationship for the MBV in the 50%-quartile with an estimate of -0.691 for the standardized variables, but we do not find a general trend if higher leverage predicts a higher significance or estimate for the MBV. Hence, we cannot confirm the impact of leverage on the correlation of MBV and tail risk as indicated by the theoretical model in section 3.

5.2.3 Results under varying size

Since size has a big impact on the coefficient and t-value of MBV if added to the regression (see table 4), we further study the relationship between size and MBV by regressing after dividing the sample into size quartiles as we did with leverage before. We run the regressions with MBV as well as with MBV and size to further explore the relations between size and MBV on tail risk.

Table 8: This table presents results from a set of OLS regressions that estimate the relationship of MBV (and size) on tail risk, after dividing the sample into quartiles depending on their size:

$$return_{i}^{crisis} = \beta_{0} + \beta_{1} * MBV_{i}^{pre-crisis} + \sum_{i} \beta_{2} * size_{i}^{pre-crisis}$$

All variables are standardized to have a mean of zero and a standard deviation of one.

	Dependent variable:									
	Return on bad bank days/ tail risk									
	25%- 50%- 75%- 100%- 25%- 50%- 75%- 100%-									
	Quartile	Quartile	Quartile	Quartile	Quartile	Quartile	Quartile	Quartile		
MBV	0.129	-0.357***	-0.485***	-0.106	0.131	-0.342***	-0.380***	-0.008		
	(1.322)	(-5.049)	(-6.787)	(-1.743)	(1.289)	(-4.858)	(-5.46)	(-0.132)		
	[0.015]	[0.187]	[0.293]	[0.026]	[0.015]	[0.177]	[0.213]	[0.000]		
Size					-0.014	-0.779	-1.337***	-0.361***		
					(-0.076)	(-1.884)	(-4.597)	(-4.561)		
					[0.000]	[0.031]	[0.161]	[0.158]		
Intercept	0.917***	0.442^{***}	-0.197***	-1.024***	0.904***	0.104	-0.083	-0.621***		
	(11.854)	(8.428)	(-3.469)	(-11.412)	(5.028)	(0.554)	(-1.434)	(-5.125)		
Observations	114	113	113	114	114	113	113	114		
\mathbb{R}^2	0.015	0.187	0.293	0.026	0.015	0.212	0.407	0.180		
Adjusted R ²	0.007	0.179	0.287	0.018	-0.002	0.198	0.396	0.165		
Note:					p-values	: *p<0.05 '	**p<0.01	**p<0.001		

Regression Results:	Varying size
----------------------------	--------------

t-values in (), partial r^2 in []

In table 8, we find that MBV seems to have the most explanatory power in both estimates and t-values for the middle quartiles (50% and 75%), and is much less significant for extreme size values. Interestingly, the size factor has a higher explanatory power for the higher quartiles (75% and 100%), but much less explanatory power in the lower quartiles. In the highest quartile, we observe a large drop in significance and estimate of the MBV when adding size, which is much more significant. Since MBV increase with size (Kane, 2014), which is confirmed in our own data sample (see appendix table 14), one possible hypothesis could be that for large firms, such as firms that are "too big to fail", the firm size incorporates some information otherwise given through the MBV. This would mean that size, especially for large firms, partly could be related to government guarantees, as indicated by studies of Gandhi & Lustig, 2015 and Acharya et al, 2016a.

5.2.4 Quantile regression

Lastly, we perform a robustness test by using quantile regression for the standard regression using market-to-book value. Quantile regressions incorporate estimates for both conditional mean and conditional volatilities, hence this method does not require the distributional assumptions of OLS to be an efficient estimator and serves as an additional robustness check. Also, by estimating the median and other quantiles the results are less sensitive to outliers than by estimating the mean.

We obtain the following findings:

Table 9: This table presents results from quantile regression to estimate the relationship between tail risk and MBV.

		Qua	antile regressi	ion						
	Dependent variable:									
-		Ret	urn on bad ba	nk days/ tail ri	sk					
	OLS		Que	antile regressi	on					
		(5%)	(25%)	(50%)	(75%)	(95%)				
MBV	-0.540***	-0.421***	-0.689***	-0.721***	-0.488***	-0.386***				
	(-13.876)	(-4.005)	(-11.239)	(-11.134)	(-8.507)	(-9.923)				
Intercept	-0.000	-1.580***	-0.528***	0.076	0.629***	1.179^{***}				
	(0.000)	(-13.376)	(-8.67)	(1.416)	(13.742)	(22.030)				
Observations	470	470	470	470	470	470				

All variables are standardized to have a mean of zero and a standard deviation of one.

--

Note: t-values in (p-values: *p<0.05 **p<0.01 ****p<0.001

t-values in ()

In table 9, we find significant (0.1% level) and negative estimates for the market-to-book values in all regressions. We see that the regression has higher significance in the median and 25%-quantile than in the peripheral quantiles. In conclusion we find that regression is not only significant and negative in the mean but also in other quantiles.

Overall, our robustness test show that MBV is a significant factor after accounting for other control variables, different leverage levels or different quantiles. Regardless of which control variables we include, MBV has a significant impact on tail risk. This is a strong sign that MBV has an explanatory power that does not come from its correlation with other relevant variables. Besides MBV, we find highly significant control variables in size, book as well as market

leverage, intangible assets, deposits and beta, with especially high coefficients and t-values for beta and size. As further robustness test, we re-sample the original data according to leverage and size, finding that while MBV as a factor is still significant across different leverage levels, MBV seems to be less significant when looking at the lowest and the highest size quartile.

We also find that our findings are significant by using quantile regression, estimating other quantiles than the mean.

As an interesting finding, we find an R^2 of over 70% when adding all control variables, which indicates that other bank ratios can also explain part of the variation in tail risk. In our next step, we try to use these other variables, besides MBV, to develop a model allowing us to estimate tail risks in an out-of-sample application.

5.3 Estimation of tail risk

We have seen that MBV and several other variables show significant correlation with banks' tail risk. In this final part, we try to make a comparison about the estimated tail risk measures in 2006 and 2016. Hence, we will use these findings to build a model with the estimates for the recent financial crisis to estimate the current tail risk banks are exposed to.

Unlike before we will not standardize the variables. First, we again run the OLS regression in the following style:

$$return_{i}^{crisis} = \beta_{0} + \sum \beta_{j} * variable_{i}^{pre-crisis}$$

Since including a large amount of variables leads to higher variance and could lead to overfitting, we apply two methods for model selection: machine learning (LASSO) and model selection using AIC in a stepwise algorithm. The least absolute shrinkage and selection operator (LASSO) is a regression analysis method that can be applied to variable selection by introducing a penalty-term λ for each included variable (Tibshirani, 1996). We can apply cross-validation to find the optimal penalty-term λ , which is a method aiming to minimize the residual sum of squares.

Our second method model selection using the Akaike Information Criterion (AIC) aims at minimizing information loss (Akaike, 1973), by adding variables to the existing model if the combined model shows a lower AIC than the previous model.

Since both models suggest a different number of variables, they produce different results. Table 9 shows the coefficients, t-values and partial R^2 of the coefficients of both models. We do not

standardize the coefficients for these regressions, as we apply these to a different sample later. Similar to tables (4) - (6), the most significant variables are still MBV, size and ROE.

To differentiate between the models, we include some metrics for comparing both models in table 10. We find that the AIC model is a better model in terms of AIC, R^2 , adj. R^2 , SSR than the LASSO model. However, the LASSO model is better in terms of BIC and has less variables. Hence, we will use both models for the estimation.

Table 10: This table presents both models from the AIC and LASSO model selection, after been chosen in the following style:

$$return_{i}^{crisis} = \beta_{0} + \sum \beta_{j} * variable_{i}^{pre-crisis}$$

Reį	gression Results: A	AIC vs LASSO model				
	Dependent variable:					
	Return o	n bad bank days/ tail risk				
	Lasso-Model	AIC-Model				
MBV	-0.003***	-0.005***				
	(-3.994)	(-5.224)				
	[0.034]	[0.058]				
Size	-0.008***	-0.008***				
	(-17.713)	(-16.403)				
	[0.412]	[0.377]				
Non-Interest	0.242^{**}	0.157^{*}				
Income	(3.249)	(2.009)				
	[0.023]	[0.009]				
ROE		0.044^{***}				
		(3.387)				
		[0.025]				
Interest Income	-0.168*	-0.156				
	(-2.2)	(-1.576)				
	[0.011]	[0.006]				
Loans/	-0.010^{*}	-0.009				
Assets	(-2.061)	(-1.836)				
	[0.009]	[0.008]				
Intangibles/	-0.054^{*}	-0.068*				
Assets	(-2.049)	(2.538)				
	[0.009]	[0.014]				
ROA		-0.180				
		(-1.63)				
		[0.006]				
Intercept	0.060^{***}	0.066^{***}				
	(11.816)	(12.072)				
Observations	454	454				
\mathbb{R}^2	0.612	0.622				
Adjusted R ²	0.607	0.615				
Note:		p-values: *p<0.05 **p<0.01 ***p<0.001				

Regression Results: AIC vs LASSO model

t-values in (), partial r² in []

Table 11: This table presents the different comparison metrics: Akaike Information Criterion, Bayesian Information Criterion, coefficient of determination (R^2), adjusted coefficient of determination (adj. R^2) and the sum of squared residuals for the LASSO and the AIC model.

	AIC	BIC	RSQ	RSQ adj	SSR
lasso_model	-2883,18	-2850,23	0,612	0,607	0,045
AIC_model	-2890,75	-2849,57	0,622	0,615	0,044

The first important observation is the relatively high R^2 of over 60%. This means that with the measures we have identified to be relevant for tail risk, in 2006 we can explain more than half of the variation in losses on bad days during the financial crisis. We test the model fit by testing the in-sample fit and we plot the actual response values (tail risk) against the predicted models (constructed tail risk) in Figure 1. We find a close approximation of the values towards the bisecting line in both plots, which shows that the predicted values are very close to the actual values of tail risk.

Figure 1: This figure plots the 2006 predicted tail risk variable against the 2006 measured tail risk variable for the AIC model and the LASSO model.



In the next step, we estimate the "potential tail risk" in 2016¹. Since the firms used in the original regression form 2006 do not exactly match the firms from 2016 due to bankruptcy or M&A, we only find 209 firms in 2016, which still exist under the same PERMCO than in 2006. We compare the estimation of the AIC and the LASSO model of these firms, which stayed the same during both 2006 and 2016, in figure 2.

¹ We take the values from 2016 to have a greater size of firms. We have a smaller sample size in 2017 since some firms might not yet have filled in their reports.

Figure 2: This figure shows the scatterplot for the actual tail risk measure in 2016 and predicted tail risk measure in 2006 and 2016. On the left side is the prediction using the LASSO-model, on the right side is the prediction using the AIC-model. The figure entails the 209 firms which operated and filled in their SEC reports in both 2006 and 2016.



Since this sample suffers from survivorship bias and has much less observations than the original regression, we add all available firms in 2006 and 2016 to the plot in figure 2 for a view on the whole banking sector, giving us 454 observations in 2006 and 325 observations in 2016.

Figure 3: This figure shows the scatterplot for the actual tail risk measure in 2016 and estimated tail risk measure in 2006 and 2016. On the left side is the prediction using the LASSO-model, on the right side is the estimation using the AIC-model. The figures entail the 454 firms for 2006 and 325 firms for 2016.



For robustness, we plot the predicted values for 2006 and 2016 of all available firms (figure 3). First, we do not estimate higher or lower tail risk measures for firms that disappeared in the financial crisis or after, compared to the ones still existing. Second, even after including firms which newly emerged, we do not find a specific difference in 2006 and 2016 tail risk estimations. Additionally, we plot the histograms for the estimated values for tail risk in 2006 vs. 2016 with both models in figure 7 (see appendix). We find a slightly more stable histogram with less fat tails in 2016 compared to 2006. However, this could also be due to the fact that in 2016, we have less firms.

This means that the potential tail risk posed by the individual banks would be still high in 2016, when estimated using our model with market and accounting data. Since we cannot compare the estimates of our model, as they are not standardized, we look into the summary statistics of the 2006 (appendix table 15) vs. the 2016 (appendix table 16) sample for comparison. We see a decrease in the mean of the MBV from 2.31 to 1.9 from the 2006 to the 2016 sample. Also, we observe a decrease in the mean value of ROE. However, another significant value is the size factor, which increased in its mean from 7.34 to 8.38 from 2006 to 2016. Interestingly, our findings observe little change in the tail risk of the individual banks. Whereas some variables like ROE and MBV decreased in its mean, leading to lower tail risk, other variables like size increased in its mean, leading to higher tail risk.

Applying our findings from the section 3, this would mean that banks in 2016 rely less on future government guarantees that are priced by investors. The increase in the size of banks, however, leaves questions open. The higher firm size compared to 2006 results in risk levels in 2016 similar to before the crisis, even though MBV decreased on average.

Although we find no or little changes in tail risk of banks in their pre- vs. post-crisis levels (consistent with findings of Sarin & Summers, 2016), we are hesitant in interpretation of our tail risk estimation as this application faces many limitations due to its many assumptions. The most prominent one is the assumption that the estimates from the 2006 regression do not change in their explanatory power until 2016. There might be different possible qualitative explanations why market behavior should have changed, which might have an effect on the relationship of MBV and other variables on tail risk. Further research in this field could increase our understanding.

To sum up, we have seen that we can explain more than 60% of the variation in returns on bad days in the recent financial crisis using the most significant measures out of the relevant

measures identified in section 5.2. Even if this prediction again confirms the importance of MBV for tail risk, it shows that we also have to take into account other measures, especially the size of the firm. Looking at our predicted levels for tail risk, we find that, based on the levels of 2016, banks would still experience similar losses on the worst days, given a crisis happened after 2016, which is similar to the one that started in 2007.

6. Discussion and conclusion

The goal of this thesis was to investigate whether market participants considered possible government guarantees when valuing banks' market values before the recent financial crisis and, if yes, how this helps us to get information about the current exposure of certain institutions to systemic risk. Using a theoretical model, we showed that the market-to-book value of a bank should be higher the higher its tail risk, given anticipated government guarantees are priced in the market value of equity. If such bailouts guarantee a minimum return of the risk-free rate to debtholders in case of a systemic breakdown, they lead to an increase in expected return of risky projects and, hence, to an increase in banks' MBV. Therefore, such projects are executed more often, leading to higher returns in normal times, but also to higher losses in bad times.

We use this model to test whether anticipated government guarantees were priced in the market value of equity in 2006. We do this running several regressions. First of all, we investigate the following hypothesis:

H1: There is a negative relationship between the market-to-book value in normal times and returns on bad days.

We show a highly significant negative correlation between banks' MBV in 2006 and their losses on the worst days during the crisis (as a proxy for tail risk). Especially mentionable is the relatively high R^2 of over 29%. This could be interpreted as a sign that market participants have considered anticipated government guarantees in banks' market value of equity in 2006.

Since market-to-book value can be decomposed into ROE and PE-ratio, we strengthen our result by testing the following hypotheses:

- *H2a: Return on equity does not explain any variation in tail risk that cannot be explained with the market-to-book value.*
- *H2b: Price-earnings-ratio does not explain any variation in tail risk that cannot be explained with the market-to-book value.*
- *H3: Return on equity and price-earnings ratio together do not contain any information that is not contained in the market-to-book value.*

Even though we also find a significant correlation between banks' ROE, PE-ratio in 2006 and their losses during the crisis, the variables explain less in variation and become insignificant if

regressed on tail risk together with MBV. Hence, the size of market-to-book value seems to be an indicator for a bank's exposure to systemic risk.

In order to further investigate the explanatory power of MBV with respect to tail risk, we looked at other variables being mentioned in literature as indicators for banks' systemic risk exposure. Overall, our robustness tests show that MBV is a significant factor after accounting for other control variables, different leverage/size levels or the estimation of different quantiles. Still, the estimate of MBV seems to be less significant for especially small or large firms.

However, other variables, next to MBV, seem to be significantly correlated with banks' tail risk as well. We find that size, leverage, deposits and intangibles are significant indicators for an institution's exposure to systemic risk. In our final part, we try to estimate tail risk levels in 2016 using a fitted model from 2006, which gives similar levels of tail risk than in the original sample, under the condition that using the same estimates is reasonable.

Since tail risk cannot be measured but only proxied, our approach faces limitations since the measure depends on the reliability of the marginal expected shortfall to proxy tail risk. Also, from looking only at one crisis in the US, our results might not be robust. By looking at the MBV as a contributor of risk in the banking sector using other prevailing methods and by testing it in other markets, the results could be strengthened. Our approach using market valuebased ratios to estimate risk of the banking sector faces more limitations, since it relies on the investors' beliefs about future cash flows. As an estimator, it can only measure potential tail risk under the assumption that investors keep pricing the MBV according to tail risk, i.e. closely related to the "efficient market assumption". Our estimation also faces more constraints, since we find that even though MBV decreased from 2006 to 2016, an increase in other factors, such as size, might still keep tail risk levels high. The relationship of these other significant variables on tail risk and if this is connected to anticipated government guarantees could be an interesting expansion of our thesis. Furthermore, our assumption for estimating the out-of-sample tail risk is, that the contribution of every variable, i.e. estimate, is the same in 2016 as in 2006, which might not hold. Lastly, we cannot show with certainty that the correlation of MBV with tail risk in our empirical study is attributable to anticipated government guarantees, which the findings from the theoretical model imply.

Our results show, once again, that the financial sector behaves differently than other industries, since our findings about the MBV have different implications than the ones of Fama & French (1992) and Zhang (2005), for the non-financial sector. Furthermore, our findings have

interesting implications on previous literature. Existing research on the decrease of the MBV in the banking sector after the financial crisis concluded that heavy regulation through increased capital requirements lead to a damage in the business models of banks (Sarin & Summers, 2016; Chousakos & Gorton, 2017). With our findings, however, we imply that the high MBV can be attributed to high risk levels banks were taking before the financial crisis. Hence, a decrease in the MBV after the financial crisis due to regulations could be a signal for lower systemic risk exposure, rather than a weakening of business models. A possible hypothesis would be that pre-crisis MBV mainly reflected risk-taking behavior through incorporating implicit government guarantees (Atkeson et al, 2018), which might be less pronounced after the crisis.

7. References

Acharya, V. V., Anginer, D., and Warburton, A, J., 2016a, The End of Market Discipline?, Working Paper.

Acharya, V. V., and Franks, J., 2008, Capital Budgeting at Banks: The Role of Government Guarantees, The Stern Stewart Institute, Retrieved from https://www.sternstewartinstitute.com/files/p2_03_-_capital_budgeting_at_banks.pdf

Acharya, V.V., Le, H., and Shin, H.S., 2017, Bank capital and dividend externalities, Review of Financial Studies 30(3), 988-1018.

Acharya, V. V., Pedersen, L., Philippon, T., and Richardson, M., 2016b, Measuring Systemic Risk, Review Of Financial Studies 30(1), 2-47.

Acharya, V. V., Thakor, A.V., 2016, The dark side of liquidity creation: Leverage and systemic risk, Journal of Financial Intermediation 28, 4-21.

Adrian, T., and Brunnermeier, M., 2016, CoVaR, American Economic Review 106(7), 1705-1741.

Adrian, T., and Shin, H., 2010, Liquidity and leverage, Journal Of Financial Intermediation 19(3), 418-437.

Adrian, T., Friedman, E., and Muir, T., 2015, The Cost of Capital of the Financial Sector, Staff Reports no. 755, Federal Reserve Bank of New York.

Akaike, H., 1973, Information Theory and an Extension of the Maximum Likelihood Principle, in B.N. Petrov and F. Csaki, ed.: 2nd International Symposium on Information Theory, 267-81 (Akademiai Kiado, Budapest).

Allen, F., Carletti, E., Goldstein, I., and Leonello, A., 2018, Government guarantees and financial stability, Journal Of Economic Theory 177, 518-557.

Allen, L., and Saunders, A., 2004, The Path to Basel II Adoption, Journal Of Financial Services Research 26(2), 101-102.

Atkeson, A. G., d'Avernas, A., Eisfeldt, A. L., and Weill, P.O., 2018, The Profitability and Valuation of Banks: Is it due to Banks' Franchise or Government Support?, Working Paper, UCLA.

Ball, R., 1978, Anomalies In Relationships Between Securities' Yields And Yield-Surrogates, Journal of Financial Economics 6, 103-26.

Basel Committee on Banking Supervision, 2003, The New Basel Capital Accord, Bank for International Settlements, Retrieved from https://www.bis.org/bcbs/bcbscp3.htm

Basel Committee on Banking Supervision, 2014, Fundamental review of the trading book: A revised market risk framework, Bank for International Settlements, Retrieved from http://www.bis.org/publ/bcbs265.pdf

Begley, T. A., Purnanandam, A. K., and Zheng, K.C., 2017, The Strategic Under-Reporting of Bank Risk, The Review of Financial Studies 30(10), 3376–3415.

Behn, M., Haselmann, R. F. H., and Vig, V., 2016, The Limits of Model-based Regulation, ECB Working Paper No. 1928, Retrieved from https://www.ecb.europa.eu/pub/pdf/scpwps/ecbwp1928.en.pdf?17ae15d416e9a8ff8b16bbd3c 746c471

Black, F., 1972, Capital Market Equilibrium with Restricted Borrowing, The Journal Of Business 45(3), 444-455.

Bogdanova, B., Fender, I., and Takáts, E., 2018, The ABCs of bank PBRs: What drives bank price-to-book ratios?, BIS Quarterly Review, Bank for International Settlements, Retrieved from https://www.bis.org/publ/qtrpdf/r_qt1803h.htm

Brunnermeier, M., Dong, G., and Palia, D., 2012, Banks' Non-Interest Income and Systemic Risk, AFA 2012 Chicago Meetings Paper, American Finance Association 2012 Meetings, Retrieved from https://www.princeton.edu/~markus/research/papers/CoVaR_non_interest_income.pdf

Calomiris, C., 1990, Is Deposit Insurance Necessary? A Historical Perspective, The Journal Of Economic History 50(02), 283-295.

Calomiris, C. W., and Nissim, D., 2014, Crisis-related shifts in the market valuation of banking activities, Journal of Financial Intermediation 23 (3), 400-435.

Campbell, J., and Shiller, R, 1998, Valuation Ratios and the Long-Run Stock Market Outlook, The Journal Of Portfolio Management 24(2), 11-26.

Caporale, T., 2012, Time varying CAPM betas and banking sector risk, Economics Letters 115(2), 293-295.

Chan, L.K.C., Jegadeesh, N., and Lakonishok, J., 1995, Evaluating the performance of value versus glamour stocks – The impact of selection bias, Journal of Financial Economics 38(3), 269-296.

Chava, S., and Purnanandam, A., 2011, The effect of banking crisis on bank-dependent borrowers, Journal of Financial Economics 99(1), 116-135.

Chousakos, K. T., and Gorton, G. B., 2017, Bank health post-crisis. Working Paper 23167, NBER.

Collyns, C., and Senhadji, S. A., 2002, Lending Booms, Real Estate Bubbles and the Asian Crisis, IMF Working Papers 02(20), 1-46.

Cragg, J. G., and Malkiel, B. G., 1982, Expectations and the Structure of Share Prices (University of Chicago Press, Chicago).

De Bandt, O., and Hartmann, P., 2000, Systemic Risk: A Survey, ECB Working Paper no. 35. Retrieved from https://www.researchgate.net/publication/4888720_Systemic_Risk_A_Survey DeAngelo, H., and Stulz, R. M., 2013, Why High Leverage is Optimal for Banks, NBER Working Papers 19139, National Bureau of Economic Research, Retrieved from https://corpgov.law.harvard.edu/2013/06/27/why-high-leverage-is-optimal-for-banks/

Fama, E., and French, K., 1992, The Cross-Section of Expected Stock Returns, The Journal Of Finance 47(2), 427-465.

Flannery, M., 2010, What to Do about TBTF?, 2010 Financial Markets Conference, 1-30, Federal Reserve Bank of Atlanta, Retrieved from https://www.frbatlanta.org/-/media/documents/news/conferences/2010/fmc/papers/Flannery.pdf

Gandhi, P., and Lustig, H, 2015, Size anomalies in U.S. bank stock returns, Journal of Finance 70, 733–768.

Graham, B., Dodd, D. L., and Cottle, S., (1962). Security Analysis: Principles And Technique (Mcgraw-Hill, New York).

Gropp, R., Gruendl, C., and Guettler, A., 2013, The Impact of Public Guarantees on Bank Risk-Taking: Evidence from a Natural Experiment, Review Of Finance 18(2), 457-488.

Grossman, S. J., and Hart, O.D., 1982, Corporate financial structure and managerial incentives, in John J. McCall, ed.: The Economics of Information and Uncertainty (University of Chicago Press).

Kane, E. J., 2014, Statement on measuring the funding advantages enjoyed by large, complex, and politically powerful bank holding companies, Financial Institutions and Consumer Protection Subcommittee, U.S. Senate Committee on Banking, Housing and Urban Affairs, Hearing on Debt Financing in the Domestic Financial Sector.

Keeley, M. C., 1990, Deposit insurance, risk, and market power in banking, American Economic Review 80(5), 1183-1200.

Kelly, B., Lustig, H., and Van Nieuwerburgh, S., 2011, Too-Systemic-To-Fail: What Option Markets Imply About Sector-Wide Government Guarantees, NBER Working Paper No. 17149, Retrieved from https://www.nber.org/papers/w17149

Laeven, L., Ratnovski, L., and Tong, H., 2016, Bank size, capital, and systemic risk: Some international evidence, Journal Of Banking & Finance 69, 25-34.

Lintner, J., 1965, The Valuation of Risk Assets and the Selection of Risky Investments in Stock Portfolios and Capital Budgets, The Review Of Economics And Statistics 47(1), 13-37.

Meiselman, B., Nagel, S., and Purnanandam, A., 2018, Judging Banks' Risk by the Profits They Report. Working Paper.

Modigliani, F., and Miller, M. H., 1958, The Cost of Capital, Corporation Finance and the Theory of Investment, The American Economic Review 48(3), 261-297.

Patro, D., Qi, M., and Sun, X., 2013, A simple indicator of systemic risk. Journal Of Financial Stability 9(1), 105-116.

Penman, S., 1996, The articulation of price-earnings ratio and market-to-book ratios and the evaluation of growth, Journal of Accounting Research 34(2), 235-259.

Rosenberg, B., Reid, K., Lanstein, R., 1985, Persuasive evidence of market inefficiency, Journal of Portfolio Management 11(3), 9–16.

Sarin, N., and Summers, L. H., 2016, Have big banks gotten safer?, Brookings Papers on Economic Activity, 15-16.

Saunders, A., and Walter, I., 1994, Universal Banking in United States: What Could We Gain? What Could We Lose? (Oxford University Press, New York).

Sharpe, W., 1964, Capital Asset Prices: A Theory of Market Equilibrium under Conditions of Risk, The Journal Of Finance 19(3), 425-442.

Shen, P., 2000, The P/E Ratio and Stock Market Performance, Economic Review Of The Federal Reserve Bank Of Kansas City, 23-36.

Tibshirani, R., 1996, Regression Shrinkage and Selection via the Lasso, Journal of the Royal Statistical Society, Series B (Methodological) 58(1), 267-288.

U.S. Department Of The Treasury, 2012, The Financial Crisis Response In Charts, Retrieved from https://www.treasury.gov/resource-center/data-chart-center/Documents/20120413_FinancialCrisisResponse.pdf

Vallascas, F., and Keasey, K., 2012, Bank resilience to systemic shocks and the stability of banking systems: Small is beautiful, Journal Of International Money And Finance 31(6), 1745-1776.

Varotto, S., and Zhao, L., 2018, Systemic risk and bank size, Journal of International Money and Finance 82, 45-70.

Weiß, G. N.F., Neumann, S., and Bostandzic, D., 2014, Systemic risk and bank consolidation: International evidence, Journal of Banking & Finance 40, 67-103.

Zhang, L., 2005, The Value Premium, The Journal of Finance 60, 67-103.

8. Appendix

Figure 4: The plot shows the historical development of the market-to-book value of the financial sector from Jan 1976 to Dec 2016. Data is taken from CRSP accessed through WRDS. The ratio is calculated as the sum of the market value of equity divided by the sum of the book value of equity of all banks in our sample at year end.



Figure 5: The plot shows the historical development of the average return on equity per year of the banks in our sample from Jan 1976 to Dec 2016. Data is taken from CRSP accessed through WRDS.



Figure 6: The plot shows the historical development of the average PE-ratio per year of the banks in our sample from Jan 1976 to Dec 2016. Data is taken from CRSP accessed through WRDS.



Figure 7: This figure shows the histograms for the tail risk values in 2006 and 2016 for the estimated values using the AIC and the LASSO model. The figures entail the 454 firms for 2006 and 325 firms for 2016.



Table 12: This table presents results from a set of OLS regressions that estimate the relationship between tail risk and prior accounting and market ratios with the regression model:

$$return_{i}^{crisis} = \beta_{0} + \sum_{i} \beta_{j} * variable_{i}^{pre-crisis}$$

The variable for tail risk is calculated using the average returns on the 5% worst days of the whole market. All variables are standardized to have a mean of zero and a standard deviation to one.

				Dependen	t variable:			
			Return	on bad ma	rket days/	tail risk		
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
MBV	-0.504***		-0.467***		-0.487***		-0.501***	-0.506**
	(-12.616)		(-11.395)		(-7.595)		(-12.463)	(-6.78)
	[0.254]		[0.218]		[0.11]		[0.25]	[0.009]
ROA		-0.265***	-0.142***					
		(-5.940)	(-3.459)					
		[0.07]	[0.025]					
ROE				-0.403***	-0.022			0.005
				(-9.504)	(-0.348)			(0.049)
				[0.162]	[0.000]			[0.000]
PE						0.080^{*}	0.024	0.026
						(1.732)	(0.601)	(0.006)
						[0.006]	[0.001]	[0.000]
Intercept	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Observations	469	469	469	469	469	469	469	469
\mathbf{R}^2	0.254	0.070	0.273	0.162	0.254	0.006	0.255	0.255
Adjusted R ²	0.253	0.068	0.270	0.160	0.251	0.004	0.252	0.250
Note:					p-values	: *p<0.05	**p<0.01 *	**p<0.00

Regression Results: Bad market days

t-values in (), partial r² in []

Summary statistics: logarithmic values					
Statistic	Ν	Mean	St. Dev.	Min	Max
Return on bad bank days (%)	440	-4.13	0.96	-7.65	-2.66
Equity (billions)	440	5.03	1.38	2.66	10.93
Assets (billions)	440	7.61	1.51	4.65	14.19
Return on Assets (%)	440	-3.21	0.19	-3.90	-2.53
Return on Equity (%)	440	-2.13	0.62	-4.49	-0.24
Market-to-book value	440	0.82	0.37	-0.21	2.18
Price-earnings ratio	440	2.95	0.45	1.58	4.86

Table 13: In this table we show the summary statistics after applying logarithmic transformation on the variables. We apply logarithms on all variables, after truncating positive outliers for return on bad bank days (tail risk). The rest follows table 1.

Table 14: This table presents results from a set of OLS regressions that estimate the relationship between MBV and size with the regression model, after dividing the initial sample into size quartiles:

$$MBV_i^{pre-crisis} = \beta_0 + \beta_1 * size_i^{pre-crisis}$$

All variables are standardized to have a mean of zero and a standard deviation of one.

		8				
		Depende	nt variable:			
-	MBV					
	25%-Quartile	50%-Quartile	75%-Quartile	100%-Quartile		
Size	0.473**	0.672	1.371***	0.472***		
	(2.918)	(1.214)	(3.659)	(4.06)		
	[0.071]	[0.013]	[0.108]	[0.128]		
Intercept	-0.177	-0.014	0.028	0.164		
	(-1.06)	(-0.054)	(0.352)	(0.863)		
Observations	114	113	113	114		
\mathbb{R}^2	0.071	0.013	0.108	0.128		
Adjusted R ²	0.062	0.004	0.100	0.121		
Note:			p-values: *p<0.05 **	*p<0.01 ****p<0.001		

Regression Results

t-values in (), partial r² in []

Table 15: In this table we show the summary statistics of key variables used for the estimation in 2006. The construction of the variables is explained in table 17.

Statistic	N	Mean	St. Dev.	Min	Max
Return on bad bank days (%)	454	-1.95	1.60	-6.98	1.52
ROA (in %)	454	4.02	0.74	2.02	7.93
ROE (in %)	454	12.72	7.57	1.13	78.68
MBV	454	2.31	0.96	0.81	8.84
PE	454	23.32	17.67	8.83	144.63
Size	454	7.34	1.42	4.65	14.19
Book leverage (in %)	454	14.23	8.66	0.00	49.47
Market leverage (in %)	454	42.26	19.04	0.00	87.50
Non-interest income (in %)	454	0.95	0.73	0.03	5.15
Interest income (in %)	454	5.88	0.78	3.37	9.33
Loans/Assets (in %)	454	70.51	12.23	7.95	93.62
Deposits/Assets (in %)	454	74.21	8.95	42.84	89.90
Intangibles/Assets (in %)	454	1.74	2.20	0.00	15.97
Dividends/Assets (in %)	454	4.52	3.97	0.00	38.16
Predicted tail risk (AIC)	454	-0.019	0.013	-0.073	0.005
Predicted tail risk (LASSO)	454	-0.019	0.013	-0.076	0.004

Summary statistics: 2006 AIC/ LASSO estimation

Summary statistics: 2016 AIC/ LASSO estimation					
Statistic	Ν	Mean	St. Dev.	Min	Max
ROA (in %)	325	1.71	0.84	0.39	10.77
ROE (in %)	325	10.01	4.06	1.15	27.49
MBV	325	1.90	0.74	0.002	4.85
PE	325	21.71	12.88	0.03	95.98
Size	325	8.38	1.82	5.68	14.73
Book leverage (in %)	325	9.81	6.43	0.00	47.70
Market leverage (in %)	325	35.88	19.67	0.00	99.91
Non-interest income (in %)	325	0.96	0.68	0.06	7.06
Interest income (in %)	325	3.47	0.99	1.07	14.95
Loans/Assets (in %)	325	69.50	11.93	19.03	91.15
Deposits/Assets (in %)	325	77.49	7.98	38.91	91.22
Intangibles/Assets (in %)	325	1.60	1.62	0.00	10.11
Dividends/Assets (in %)	325	3.21	2.54	0.00	17.21
Predicted tail risk (AIC)	325	-0.019	0.015	-0.068	0.011
Predicted tail risk (LASSO)	325	-0.022	0.014	-0.065	0.007

Table 16: In this table we show the summary statistics of key variables used for the estimation in 2016. The construction of the variables is explained in table 17.

Example 1: We show that in a setting with government guarantees we will obtain a MBV larger than 1 in a state with government guarantees:

(13)
$$M = \frac{1}{R^f} \times (1 - q_c) \times \left[\left(R^n - \boldsymbol{\rho} \times \boldsymbol{R}^f \right) \times I - N + M \right] > N$$

 $R^{f} = 1.02$ $R^{n} = 1.03$

 $q_c=0.03$

$$\rho = 0.8$$

By plugging these numbers in equation (13) above, we get

$$M = 1.018 \times \left[0.951 \times \frac{N}{0.1} - N + M \right]$$

And simplified:

$$M = 1.358 \times N$$

Table 17: In this table we present the computation of the variables. We obtain all variables from the Wharton Research Database Services (WRDS) from the CRSP for daily stock market data and the joined CRSP/Compustat database for bank annual data.

Variable name	Definition	Data description (CRSP, Compustat)
MBV	Market capitalization over tangible book value of equity	$\frac{prc \times shrout}{ceqt}$
ROE	Net income over tangible book value of equity	ni ceqt
ROA	Earnings before interest expense over tangible book value of equity	$\frac{ni - xint + txt}{ceqt}$
PE	Market capitalization over net income	$\frac{prc \times shrout}{ni}$
Size	Log Assets	ln (at)
Book leverage	Debt (current + long-term) over assets	$\frac{dltt + dlc}{at}$
Market leverage	Debt (current + long-term) over debt + market capitalization	$\frac{dltt + dlc}{dltt + dlc + prc \times shrout}$
Non-Interest income	Non-interest income over assets	tnii at
Interest income	Interest income over assets	$\frac{idit}{at}$
Intangibles/Assets	Intangibles over assets	$\frac{intan}{at}$
Loans/Assets	Loans over assets	$\frac{lg}{at}$
Deposits/Assets	Deposits over assets	$\frac{dptc}{at}$
Dividends/Equity	Dividends over tangible book value of equity	$\frac{dvc}{ceqt}$
Beta	Covariance (stock, market) over variance of the market	$\frac{cov(i,m)}{var(m)}$